

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Chen, Natalie; Novy, Dennis

Working Paper International trade integration: a disaggregated approach

CESifo Working Paper, No. 2595

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Chen, Natalie; Novy, Dennis (2009) : International trade integration: a disaggregated approach, CESifo Working Paper, No. 2595, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/26640

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

International Trade Integration: A Disaggregated Approach

NATALIE CHEN DENNIS NOVY

CESIFO WORKING PAPER NO. 2595 CATEGORY 8: TRADE POLICY MARCH 2009

PRESENTED AT CESIFO AREA CONFERENCE ON GLOBAL ECONOMY, FEBRUARY 2009

An electronic version of the paper may be downloaded • from the SSRN website: www.SSRN.com • from the RePEc website: www.RePEc.org • from the CESifo website: www.CESifo-group.org/wp

International Trade Integration: A Disaggregated Approach

Abstract

This paper investigates the sources and size of trade barriers at the industry level. We derive a micro-founded measure of industry-specific bilateral trade integration that has an in-built control for time-varying multilateral resistance. This trade integration measure is consistent with a broad range of recent trade models including the Anderson and van Wincoop (2003) framework, the Ricardian model by Eaton and Kortum (2002) and heterogeneous firms models. We use it to explore trade barriers for manufacturing industries in European Union countries between 1999 and 2003. We find a large degree of trade cost heterogeneity across industries. The most important trade barriers are transportation costs and policy factors such as Technical Barriers to Trade. Trade integration is generally lower for countries that opted out of the Euro or did not abolish border controls in accordance with the Schengen Agreement. Reductions in trade barriers explain about one-half of the growth in trade over the period 1999-2003 and are therefore a major driving force of the EU Single Market.

JEL Code: F10, F15.

Keywords: trade integration, gravity, trade costs, multilateral resistance, industries, disaggregation, European Union.

Natalie Chen Department of Economics University of Warwick UK - Coventry CV4 7AL N.A.Chen@warwick.ac.uk Dennis Novy Department of Economics University of Warwick UK - Coventry CV4 7AL D.Novy@warwick.ac.uk

March 12, 2009

We are grateful for comments by Richard Baldwin, Pierre-Philippe Combes, Donald Davis, Ronald Findlay, James Harrigan, Thierry Mayer, Steve Redding, David Weinstein and participants at the CEPR ERWIT 2007, Midwest Spring Meetings 2008, the Product Market Integration in the EU 2008 workshop at the Universidade do Minho, Columbia, Fordham, Georgia Tech, HEI Geneva, Nottingham, Oxford, Santa Clara, UC Davis, UC Santa Cruz, Utrecht School of Economics, Vanderbilt and Yale.

Novy gratefully acknowledges financial support from the Economic and Social Research Council, Grant RES-000-22-3112.

1 Introduction

Trade costs are a staple ingredient in today's trade literature. Broadly defined, trade costs include any cost of engaging in international trade such as transportation costs, tariffs, non-tariff barriers, informational costs, time costs, different product standards, exchange rate costs and local distribution costs, among others. They feature prominently in the vast majority of theoretical papers, including the models of Eaton and Kortum (2002) and Melitz (2003). In addition, a growing empirical literature, surveyed by Anderson and van Wincoop (2004), is devoted to exploring the sources and size of trade costs. A deeper understanding of the causes of trade costs is of particular importance because it would enable a better evaluation of their welfare implications. These are suspected to be large: on their own, policy-related trade costs may be worth more than ten percent of national income (Anderson and van Wincoop, 2002).

A major challenge faced by empirical researchers is to measure overall trade costs since "direct measures are remarkably sparse and inaccurate" (Anderson and van Wincoop, 2004, p.692). Direct measures are only available for a few components, for instance transportation and insurance costs, usually proxied by the ratio of *c.i.f.* and *f.o.b.* trade values (Harrigan, 1993, Hummels, 2001a, 2007),¹ policy barriers such as specific tariff or non-tariff barriers (Chen, 2004, Harrigan, 1993, Head and Mayer, 2000), informational costs (Rauch, 1999) or time costs (Evans and Harrigan, 2005, Harrigan, 2005, Hummels, 2001b). But even for those components, data coverage is often limited to a few countries and years, and it can be hard to gather disaggregated trade cost data at the industry or product level.

Given those difficulties in obtaining accurate measures of trade costs, some researchers indirectly infer the level of trade impediments from trade flows. This approach has the obvious advantage of extending the analysis to more countries, years and more finely disaggregated data. One way of doing this is to compute trade to output ratios (Harrigan, 1996); another is to estimate "border effects," which mostly reflect the extent of border-related costs.²

This paper is part of the research effort that attempts to indirectly infer trade impediments from trade flows. The first contribution of the paper is to develop a micro-founded measure of bilateral trade integration that can be applied to disaggregated panel data. We derive this measure by modeling disaggregated trade flows at the industry level in the gravity framework pioneered by Anderson and van Wincoop (2003, 2004), allowing trade costs and substitution elasticities to be heterogeneous across industries. Anderson and van Wincoop (2003) show that trade flows are determined not only by bilateral trade costs between two countries but also by average trade barriers with other countries, which they refer to as "multilateral resistance." Following the approach of Novy (2008), we derive an analytical solution for multilateral resistance variables that vary across industries and over time. In turn, this enables us to derive a micro-founded measure of bilateral trade integration that has an in-built control for time-varying multilateral resistance and can therefore be applied to panel data.

Arguably, the Anderson and van Wincoop model is one of the most parsimonious trade models of recent years. It rests on the Armington assumption that countries produce differentiated goods and

¹Moreover, Limão and Venables (2001) use the quotes from shipping firms for a standard container shipped from Baltimore to several destinations. Combes and Lafourcade (2005) develop a new methodology to compute transportation costs and apply it to road transport by truck in France.

²Examples include Anderson and van Wincoop (2003), Baldwin, Forslid, Martin, Ottaviano and Robert-Nicoud (2003), Chen (2004), Eaton and Kortum (2002), Evans (2003), Head and Mayer (2000), Head and Ries (2001), McCallum (1995), Nitsch (2000) and Wei (1996).

trade is driven by consumers' love of variety, leading to the key gravity equation. However, we show that isomorphic trade integration measures can be derived from the Ricardian trade model by Eaton and Kortum (2002), from Chaney's (2008) extension of the Melitz (2003) heterogeneous firms model as well as from the heterogeneous firms model by Melitz and Ottaviano (2008) with linear non-CES demand. This is possible because all these models lead to gravity equations that have a similar structure. Our trade integration measure is therefore consistent with a broad range of the recent theoretical trade literature and is not limited to any one particular framework.

Although we regard our approach as complementary to other research that indirectly infers trade impediments from trade flows, we discuss how our trade integration measure differs from ad hoc proxies commonly used in the literature such as trade to output ratios. In particular, we show that ad hoc proxies without micro-foundations can erroneously pick up changes in *multilateral* trade barriers as changes in *bilateral* trade barriers. We are instead able to focus on bilateral trade integration because our micro-founded measure is not distorted by multilateral resistance effects.

The second contribution of the paper is to bring our measure of trade integration to the data. This enables us to document and explain the variation of trade barriers across 166 manufacturing industries in 11 European Union countries over the period 1999-2003. The case of the European Union is appealing since trade integration is expected to be strong between its member states due to two factors. First, these countries have succeeded in dismantling many restrictions on trade, including tariffs and quotas that were completely eliminated by 1968. Second, the situation has been further reinforced by the implementation of the Single Market Programme (SMP), launched in the mid-1980s.

Consistent with the standard gravity literature, the variation of trade integration across country pairs can to a large extent be captured by typical gravity variables such as distance and adjacency but also by policy-related variables such as membership in the Eurozone or participation in the Schengen Agreement. But our focus lies on the substantial degree of heterogeneity in trade integration across industries. Our results suggest that modeling trade costs as a "one-fits-all" impediment is clearly at odds with empirical evidence.

We investigate the role of several sectoral characteristics in explaining trade integration across industries, with a particular emphasis on policy-related variables such as the extent of Technical Barriers to Trade (TBTs). Such barriers are a predominant concern in today's global trade negotiations, and for the WTO in particular as it precisely seeks to ensure that "technical regulations and standards, including packaging, marking and labelling requirements [...] do not create unnecessary obstacles to international trade."³ We find that trade integration is indeed lower in countries and industries where TBTs are strong, suggesting that there is room left for policy action and that further gains are possible through the reduction of those barriers. We also show that trade integration tends to be high for industries characterized by high productivity, low transportation costs and a high degree of transparency in public procurement. From a dynamic perspective, average trade integration has improved for most countries over the period 1999-2003, as well as individually for a large number of industries in our sample.

We also contrast our empirical approach of explaining the variation in bilateral trade integration to the standard gravity approach of regressing bilateral trade flows. We believe this is an important exercise that yields two insights. First, our approach allows us to identify the effects of a number of crucial

 $^{^{3}}$ Agreement on Technical Barriers to Trade (p.117). This Agreement, negotiated during the Uruguay Round, is an integral part of the WTO Agreement.

explanatory variables such as weight-to-value that would normally drop out in a correctly specified gravity equation due to perfect collinearity with fixed effects. Second, we show that misspecifying a standard gravity equation by omitting multilateral resistance variables leads to a substantial bias in the estimated trade cost elasticities. Our approach avoids this problem because our trade integration measure has an in-built control for time-varying multilateral resistance.

The third contribution of the paper is to explore to what extent the increase in trade flows over recent years can be explained by the decrease in trade barriers. For that purpose, we use our model to decompose the growth in trade into two main components -(1) the growth of manufacturing output and (2) improvements in trade integration. We find that on average 58 percent of the growth in trade can be accounted for by changes in bilateral trade barriers and multilateral resistance, while the rest can be attributed to the secular growth of manufacturing output. This contribution of improved trade integration is quantitatively larger than the contribution found by Baier and Bergstrand (2001) for a sample of OECD countries. As we focus on the European Union, our results suggest that on average, intra-EU trade integration has progressed faster than integration amongst other developed countries.

The paper is organized as follows. In Section 2 we develop a general equilibrium model with industry-specific trade costs. In that section we also derive an analytical solution for time-varying industry-level multilateral resistance variables and our trade integration measure. Section 3 presents our dataset. Section 4 reports our empirical results, focusing on the determinants and time trend of trade integration. We also present the decomposition of trade growth into output growth and improvements in trade integration. Section 5 provides robustness checks and Section 6 concludes.

2 A Model with Industry-Specific Trade Costs

Our model closely follows the seminal paper by Anderson and van Wincoop (2003). Their general equilibrium model of trade results in a gravity equation that incorporates trade costs. The key insight from their gravity equation is that bilateral trade flows are not determined solely by the absolute bilateral trade barrier between two countries but rather by their bilateral *relative* to their average trade barrier. Anderson and van Wincoop (2003) refer to the appropriate average trade barrier as "multilateral resistance."

As a generalization, Anderson and van Wincoop (2004) model bilateral trade for an individual industry that is characterized by industry-specific bilateral trade costs and an industry-specific elasticity of substitution. We follow Anderson and van Wincoop (2004) in modeling trade flows with heterogeneous trade costs and heterogeneous elasticities of substitution at the industry level. The innovation of our approach is to derive an analytical solution for time-varying industry-specific multilateral resistance variables that can be related to observable data. With this solution at hand, we are able to derive a micro-founded measure of industry-specific bilateral trade integration that nets out multilateral resistance effects. Bergstrand (1989, 1990) also derives gravity equations for industry-level trade flows but does not focus on multilateral resistance.

2.1 The Basic Framework

Denote x_{ij}^k as nominal exports from country *i* to country *j* in goods associated with industry *k*. Suppose that consumers in country *j* allocate expenditure x_i^k on industry-*k* goods and that their preferences

over these goods can be described by a standard CES utility function as

$$C_j^k \equiv \left(\sum_{i=1}^J \left(c_{ij}^k\right)^{\frac{\sigma_k - 1}{\sigma_k}}\right)^{\frac{\sigma_k}{\sigma_k - 1}} \tag{1}$$

where c_{ij}^k is real consumption of industry-k goods from country *i* by country-*j* consumers and where the elasticity of substitution σ_k is specific to industry k and assumed to exceed unity, $\sigma_k > 1$. Furthermore suppose that the factory gate price of industry-k goods from country *i* is denoted by p_i^k and that trade costs associated with the trade cost factor $t_{ij}^k \ge 1$ are incurred when these goods are shipped to country *j* such that the price faced by country-*j* consumers, denoted by p_{ij}^k , can be written as $p_{ij}^k = t_{ij}^k p_i^k$. The demand function for exports $x_{ij}^k = p_{ij}^k c_{ij}^k$ then follows as

$$x_{ij}^{k} = \left(\frac{p_{ij}^{k}}{P_{j}^{k}}\right)^{1-\sigma_{k}} x_{j}^{k} = \left(\frac{t_{ij}^{k} p_{i}^{k}}{P_{j}^{k}}\right)^{1-\sigma_{k}} x_{j}^{k}$$

$$\tag{2}$$

where the price index P_j^k can be derived as

$$P_j^k = \left(\sum_{i=1}^J \left(p_{ij}^k\right)^{1-\sigma_k}\right)^{\frac{1}{1-\sigma_k}} \tag{3}$$

2.2 The Gravity Equation

Denote output of industry-k goods by country-i firms as y_i^k and impose market-clearing as

$$y_i^k = \sum_{j=1}^J x_{ij}^k \tag{4}$$

Substituting the demand function (2) into the market-clearing condition (4) and rearranging yields

$$p_i^k = \left(\sum_{j=1}^J \left(\frac{t_{ij}^k}{P_j^k}\right)^{1-\sigma_k} \frac{x_j^k}{y_i^k}\right)^{\frac{1}{\sigma_k-1}}$$
(5)

Plug equation (5) back into the demand function (2) and, as in Anderson and van Wincoop (2004), define outward multilateral resistance for industry-k goods from country i as

$$\Pi_i^k \equiv \left(\sum_{j=1}^J \left(\frac{t_{ij}^k}{P_j^k}\right)^{1-\sigma_k} \frac{x_j^k}{y^k}\right)^{\frac{1}{1-\sigma_k}} \tag{6}$$

where y^k is world output in industry k. Rearrange to arrive at a gravity equation for industry k

$$x_{ij}^{k} = \frac{y_i^{k} x_j^{k}}{y^{k}} \left(\frac{t_{ij}^{k}}{\Pi_i^{k} P_j^{k}} \right)^{1 - \sigma_k} \tag{7}$$

Trade flows x_{ij}^k depend on supply y_i^k of the k-good from country i and expenditure x_j^k for the good in country j. Large bilateral trade costs t_{ij}^k reduce bilateral trade, whereas large average outward trade

barriers of country i (i.e., large Π_i^k) and large average inward trade barriers of country j (i.e., large P_j^k) lead to more bilateral trade. Substituting the solution for p_i^k in (5) and the definition of Π_i^k in (6) into the price index (3) yields

$$P_j^k = \left(\sum_{i=1}^J \left(\frac{t_{ij}^k}{\Pi_i^k}\right)^{1-\sigma_k} \frac{y_i^k}{y^k}\right)^{\frac{1}{1-\sigma_k}}$$
(8)

which is inward multilateral resistance for industry-k goods entering country j.

2.3 Solving for Multilateral Resistance

A problem that arises in empirical work is that we do not have data for the multilateral resistance terms P_j^k and Π_i^k in gravity equation (7). The method we employ here is to solve for these terms analytically as a function of observable trade flows. We exploit the fact that multilateral resistance is related to the amount of trade a country conducts with itself (see Novy, 2008). Intuitively, if a country's trade barriers with the rest of the world are high (i.e., if the country's multilateral resistance is high), the country will trade a lot domestically.

To see this formally, use gravity equation (7) and consider domestic trade flows for industry-k goods

$$x_{ii}^k = \frac{y_i^k x_i^k}{y^k} \left(\frac{t_{ii}^k}{\Pi_i^k P_i^k}\right)^{1-\sigma_k} \tag{9}$$

where t_{ii}^k are domestic trade costs for industry-k goods, for example domestic transportation costs. Equation (9) can be solved for the product of outward and inward multilateral resistance as

$$\Pi_i^k P_i^k = \left(\frac{y_i^k x_i^k}{x_{ii}^k y^k}\right)^{\frac{1}{1-\sigma_k}} t_{ii}^k \tag{10}$$

Note that we do not impose zero domestic trade costs since the trade cost factor t_{ii}^k may exceed unity. Equation (10) implies that for given t_{ii}^k and σ_k , it is easy to measure multilateral resistance since the quantities on the right-hand side are observable.

2.4 A Micro-Founded Measure of Industry-Specific Trade Integration

The solution for multilateral resistance can be exploited to solve the model. Gravity equation (7) contains the product of outward multilateral resistance of country i and inward multilateral resistance of country j, $\prod_{i}^{k} P_{j}^{k}$, whereas equation (10) provides a solution for $\prod_{i}^{k} P_{i}^{k}$. It is therefore useful to multiply gravity equation (7) by the corresponding gravity equation for trade flows in the opposite direction, x_{ji}^{k} , to obtain a bidirectional gravity equation that contains both countries' outward and inward multilateral resistance variables. This yields

$$x_{ij}^{k} x_{ji}^{k} = \frac{y_{i}^{k} y_{j}^{k} x_{i}^{k} x_{j}^{k}}{y^{k} y^{k}} \left(\frac{t_{ij}^{k} t_{ji}^{k}}{\prod_{i}^{k} P_{i}^{k} \prod_{j}^{k} P_{j}^{k}} \right)^{1 - \sigma_{k}}$$
(11)

Substitute the solution for multilateral resistance given in equation (10) to obtain

$$x_{ij}^{k} x_{ji}^{k} = x_{ii}^{k} x_{jj}^{k} \left(\frac{t_{ij}^{k} t_{ji}^{k}}{t_{ii}^{k} t_{jj}^{k}} \right)^{1 - \sigma_{k}}$$
(12)

From equation (12) it is easy to solve for the trade cost factors as

$$\frac{t_{ij}^k t_{ji}^k}{t_{ii}^k t_{jj}^k} = \left(\frac{x_{ii}^k x_{jj}^k}{x_{ij}^k x_{ji}^k}\right)^{\frac{1}{\sigma_k - 1}} \tag{13}$$

It is only possible to infer relative trade costs, in this case bilateral trade costs $t_{ij}^k t_{ji}^k$ relative to intranational trade costs $t_{ii}^k t_{jj}^{k}$.⁴ We do not impose trade cost symmetry so that t_{ij}^k and t_{ji}^k on the left-hand side of equation (13) may be asymmetric ($t_{ij}^k \neq t_{ji}^k$). As Anderson and van Wincoop (2003, footnote 11) point out, it is problematic to infer the degree of trade barrier asymmetry from trade data because there are multiple combinations of t_{ij}^k and t_{ji}^k that can give rise to the same trade flows x_{ij}^k and x_{ji}^k . We therefore take the square root to get an expression for the *average* bilateral trade barrier. Thus, the average relative trade barrier can be expressed as

$$\theta_{ij}^{k} \equiv \left(\frac{t_{ij}^{k} t_{ji}^{k}}{t_{ii}^{k} t_{jj}^{k}}\right)^{\frac{1}{2}} = \left(\frac{x_{ii}^{k} x_{jj}^{k}}{x_{ij}^{k} x_{ji}^{k}}\right)^{\frac{1}{2(\sigma_{k}-1)}} \tag{14}$$

We interpret θ_{ij}^k as a micro-founded measure of bilateral industry-specific trade frictions, or the inverse of bilateral trade integration. The more two countries trade with each other (i.e., the higher $x_{ij}^k x_{ji}^k$), the lower is our measure of relative trade frictions ceteris paribus. Conversely, if the two countries start trading more domestically (i.e., the higher $x_{ii}^k x_{jj}^k$), the higher is our measure of relative trade frictions ceteris paribus.

For the interpretation of θ_{ij}^k it is also helpful to think of two opposite extreme cases – a frictionless world with no trade costs on the one hand, and a closed economy on the other. In a frictionless world, all trade cost factors equal unity $(t_{ij}^k = t_{ji}^k = t_{ii}^k = t_{jj}^k = 1)$ and θ_{ij}^k would be one, implying that bilateral and domestic barriers are the same. In the case approaching a closed economy, bilateral trade $x_{ij}^k x_{ji}^k$ tends towards zero and thus θ_{ij}^k tends towards infinity, implying that bilateral barriers are prohibitive relative to domestic barriers.

Last but not least, we stress that our trade integration measure is valid more generally beyond the particular framework presented above. We have derived θ_{ij}^k in equation (14) from a model based on a CES demand system in combination with the Armington assumption that goods are differentiated by country of origin. But as we show in the technical appendix, we can also derive our trade integration measure from a range of other leading trade theories. These include the Ricardian trade model by Eaton and Kortum (2002), Chaney's (2008) extension of the Melitz (2003) heterogeneous firms model to asymmetric countries as well as the heterogeneous firms model by Melitz and Ottaviano (2008) with linear non-CES demand and endogenous markups.

The reason why our trade integration measure θ_{ij}^k is consistent with a broad range of trade models is related to the fact that they all lead to gravity equations that have a similar structure as equation (7).⁵ Intuitively, the gravity equation is an expenditure equation that indicates how consumers allocate their expenditure across countries subject to trade frictions (Baldwin and Taglioni, 2006). Gravity equations arise regardless of *why* consumers want to buy goods from foreign countries. In an Armington world,

⁴On this point also see Anderson and van Wincoop (2004, p.709).

⁵Feenstra, Markusen and Rose (2001) and Evenett and Keller (2002) also show that gravity equations are consistent with various competing trade models.

consumers buy foreign goods because those goods are inherently different and consumers prefer variety. In a Ricardian world, countries produce goods according to comparative advantage and consumers buy foreign goods because they are cheaper. It turns out that the particular motivation behind foreign trade is not crucial to understand the role of trade frictions.⁶

2.5 A Comparison to Alternative Measures

We now contrast θ_{ij}^k with alternative measures of trade integration. First, compare θ_{ij}^k with a bilateral trade to output ratio, $\left(x_{ij}^k x_{ji}^k\right) / \left(y_i^k y_j^k\right)$. Various forms of such ratios have been used as measures of trade openness (Harrigan, 1996). Dividing gravity equation (11) by $y_i^k y_j^k$ and substituting equation (10) yields

$$\frac{x_{ij}^k x_{ji}^k}{y_i^k y_j^k} = \frac{x_{ii}^k x_{jj}^k}{y_i^k y_j^k} \left(\frac{t_{ij}^k t_{ji}^k}{t_{ii}^k t_{jj}^k}\right)^{1-\sigma_k}$$
(15)

Theory thus predicts that apart from the trade cost parameters, the bilateral trade to output ratio is also determined by the two countries' domestic trade shares, $\left(x_{ii}^k x_{jj}^k\right) / \left(y_i^k y_j^k\right)$.

To compare the trade to output ratio with θ_{ij}^k , take the trade cost parameters t_{ij}^k , t_{ji}^k , t_{ii}^k and t_{jj}^k as given. This pins down θ_{ij}^k in equation (14). However, the trade to output ratio can still move. For example, suppose that trade costs with a third country increase, say, t_{im}^k with $m \neq i, j$. The increase in t_{im}^k pushes up the domestic trade ratio x_{ii}^k/y_i^k and therefore also the bilateral trade to output ratio in equation (15).⁷ But this is a multilateral resistance movement and does not imply that trade integration between *i* and *j* has improved. In contrast, θ_{ij}^k does not react to multilateral resistance movements because θ_{ij}^k only reflects changes in bilateral and domestic trade costs, not changes in third-party trade costs. In conclusion, trade to output ratios are distorted by multilateral resistance effects.

Second, compare θ_{ij}^k with the "*phi*-ness" of trade, which is given by the simple ratio of bilateral to domestic trade flows and is commonly used to estimate sectoral border effects (Baldwin et al., 2003, Head and Mayer, 2004, Head and Ries, 2001).⁸ Theory shows that the border effect can arise because a high degree of substitution between domestic and imported goods may lead to a high responsiveness of trade flows even in the case of very modest trade barriers. However, in an industry with a high elasticity σ_k , consumers are so price-sensitive that a high ratio of domestic over bilateral trade does not necessarily reflect high bilateral trade barriers but rather a high degree of competition.⁹ The definition of θ_{ij}^k combines the *phi*-ness measure with an exponent that involves the industry-specific elasticity of substitution σ_k so in contrast to *phi*-ness, θ_{ij}^k is able to separate this competition effect from the trade

⁶Deardorff (1998) argues that in a Heckscher-Ohlin framework, trade frictions prevent factor price equalization so that for the large majority of goods, only one country is the lowest-cost producer. In the presence of trade frictions, trade in a Heckscher-Ohlin world therefore resembles trade in an Armington world. Grossman (1998) provides the following intuition for the gravity equation: "*Specialization* lies behind the explanatory power [of the gravity equation], and of course some degree of specialization is at the heart of any model of trade. Thus, the derivation of the gravity equation need not make reference to any particular trade model at all [...] This is true no matter what supply-side considerations give rise to the specialization, be they increasing returns to scale in a world of differentiated products, technology differences in a world of Ricardian trade, large factor endowment differences in a world of Heckscher-Ohlin trade, or (small) transport costs in a world of any type of endowment-based trade." (Emphasis in the original.)

⁷Formally, the increase in multilateral resistance follows from equation (6), implying an increase in x_{ii}^k through equation (10).

⁸Alternatively, border effects can be estimated from a gravity equation that combines domestic and international trade flows as the dependent variable (Chen, 2004, Evans, 2003, McCallum, 1996, Wei, 1996, Wolf, 2000).

⁹The standard CES markup is given by $\sigma_k/(\sigma_k-1)$ and thus inversely related to σ_k .

barrier effect. As equation (14) shows, a higher elasticity of substitution implies *lower* trade frictions θ_{ij}^k . Likewise, as we show in the technical appendix, the theoretical models by Eaton and Kortum (2002), Chaney (2008) and Melitz and Ottaviano (2008) also require that the trade ratio be scaled by an industry-specific exponent. We therefore argue that θ_{ij}^k improves on the *phi*-ness measure because it accounts for differences in market power and competition across industries.¹⁰

3 Data and Descriptive Statistics

To compute our measure of trade integration θ_{ij}^k across industries, countries and time, we need the domestic trade of countries *i* and *j* in industry *k*, x_{ii}^k and x_{jj}^k , as well as their bilateral exports, x_{ij}^k and x_{ji}^k , at time *t*. As in previous literature (for instance, Chen, 2004, Evans, 2003, Head and Mayer, 2000, Nitsch, 2000, Wei, 1996), domestic trade for country *i* is given by its gross output in industry *k*, y_i^k , minus total exports of country *i* to the rest of the world in that industry. The elasticities of substitution σ_k are taken from Hummels (2001a).¹¹ Our sample includes trade flows for 166 manufacturing industries across 11 EU countries at the 4-digit Nace rev.1 level between 1999 and 2003. The sample is balanced over time. The 11 countries are Austria, Denmark, Finland, France, Germany, Italy, Ireland, the Netherlands, Spain, Portugal and the United Kingdom. In the data appendix we provide a detailed description of the data and their sources.¹²

Our dataset comprises a total of 15,040 domestic and bilateral trade flow observations. Only five bilateral trade flows are equal to zero, which is not surprising given the huge volume of intra-industry trade in the EU.¹³ Those five cases would normally feature as missing values of θ_{ij}^k . But zero trade flows may contain valuable information as to why such low levels of trade are observed. It therefore seems more appropriate to associate these cases with large trade frictions. The approach we adopt is to replace the zeros with a value of one Euro, thus associating them with large values of θ_{ij}^k .¹⁴

Table 1 contains summary statistics for individual industries. Due to space constraints we list industries at the more aggregated 3-digit level, reporting averages of θ_{ij}^k over the lower-level 4-digit classifications as well as their maximum and minimum values. Industries are ordered by decreasing value of θ^k , which ranges from 146.91 to 1.15. We also report the average elasticities of substitution σ_k and the average weight-to-value ratios, measured as kilograms per Euro exported.

A first cursory glance at Table 1 attests the intuitive nature of the trade integration measure θ_{ij}^k . Trade integration is lowest for Bricks, followed by Cement, lime, plaster. The latter is also the industry with the largest average weight-to-value ratio (over 13 kg/Euro), indicating a very low transportability of the goods. Another related industry is Articles of concrete. Note that the geographic market for cement and concrete is very local since the perishable nature of such "wet" products constrains the distance over which they can be delivered.

¹⁰We show in Section 5 that regressions based on the *phi*-ness measure yield a considerably lower R^2 than regressions of θ_{ij}^k .

¹¹Hanson and Xiang (2004) also use the elasticities from Hummels (2001a).

¹²Due to data limitations not all possible trade flow combinations across industries and countries are available. Although it would in principle be possible to span the longer period from 1997 to 2003, this would come at the cost of losing all observations for Germany, a core EU country, because German sectoral output data are missing prior to 1999.

¹³The three industries with zero trade flows are Publishing of newspapers, Builders' carpentry and joinery and Mattresses. We consider those observations as "true zeros" because the corresponding output values are positive but exports are zero. We exclude the cases where output is zero but exports are positive as well as the cases where both output and exports are zero.

¹⁴Our treatment of the zero trade flows does not distort the overall results, see the robustness check in Section 5.

Printing and Publishing are traded very little, too, which is hardly surprising given the reliance of such products on specific languages. This finding is consistent with earlier studies showing that trade in such sectors is subdued. Harrigan (1996) shows that the volume of trade relative to output in the OECD in 1985 is the lowest in Printing and Publishing. Finally, some of the sectors with high values of θ^k belong to the food industry, for which the perishability of many goods is most likely an important deterrent to trade. For example, Fruit and vegetables are ranked in fourth place, and Other food products display the biggest individual value of θ^k equal to 1,056.

Table 2 reports the average level of θ_i for the 11 individual countries in our sample. Ireland appears as the least integrated country, followed by Spain, Denmark, Finland and Italy, whereas France, the Netherlands and Germany are the most trade integrated.

Finally, Figure 1 plots the time series evolution of θ_t , averaged across countries and industries. It is interesting to observe that despite fluctuations from one year to the next, trade frictions display a downward trend, suggesting that the countries and industries we consider have on average become more bilaterally integrated over time.

4 Empirical Results

In the first part of this section, we analyze the determinants of trade integration across country pairs, industries and years. In the second part, we explain how our approach differs from standard gravity. In the third part, we focus on the time series patterns of trade integration. In the fourth part, we decompose the growth in trade over the sample period into two components – the growth in manufacturing output and improvements in trade integration.

4.1 The Determinants of Trade Integration

We first analyze the determinants of bilateral EU trade integration. We estimate

$$\ln \theta_{ij,t}^k = \psi_t + \lambda_K + \beta \ Geo_{ij,t}^k + \zeta \ Policy_{ij,t}^k + \alpha \ Costs_{ij,t}^k + \gamma \ Controls_{ij,t}^k + \epsilon_{ij,t}^k$$
(16)

where $Geo_{ij,t}^k$ is a set of variables related to geography and transportation costs, $Policy_{ij,t}^k$ is a set of policy-related factors, $Costs_{ij,t}^k$ includes other types of costs such as fixed costs of exporting and productivity, and $Controls_{ij,t}^k$ includes variables controlling for measurement issues. Given that some of the explanatory variables vary across sectors only, we include industry fixed effects λ_K at the more aggregated level of 3-digit industries, assuming that the 4-digit groupings k are different varieties of the corresponding, more aggregated 3-digit sector K (Hummels, 2001a). The inclusion of other explanatory variables that only vary across country pairs precludes us from controlling for country pair fixed effects.¹⁵ ψ_t denotes year intercepts. β , ζ , α and γ are vectors of coefficients to be estimated and $\epsilon_{ij,t}^k$ is an error term. A higher value of θ_{ij}^k should be interpreted as a lower degree of trade integration. We estimate our baseline results with OLS. To control for possible autocorrelation in each of the individual series of the panel, robust standard errors are adjusted for clustering at the 4-digit Nace rev.1 level in each country pair (15,040/5 = 3,008 clusters). As a robustness check, we also report results obtained by Poisson Pseudo-Maximum Likelihood (PPML) estimation, see Section 5.

¹⁵As a robustness check, we show that our results hold up with country pair fixed effects, see Section 5.

Geography/Transportation Costs Variables Table 3 reports our main results. Column (1) only includes variables that are related to geography and transportation costs. Across countries, standard gravity variables perform well in explaining the variation in trade integration. Trade integration decreases with international distances D_{ij} and increases with domestic distances $D_{ii} \times D_{jj}$. It is higher between countries that share a common border Adj_{ij} and a common language $Lang_{ij}$.

We also consider proxies of transportation costs that vary across industries. First, we consider the ratio of *c.i.f.* and *f.o.b.* trade values for each industry and country pair, averaged in each year across all EU partners in order to minimize measurement error, $cfob_t^k$ (Harrigan, 1993).¹⁶ Second, we consider the weight-to-value ratio of exports for each industry and country pair, also averaged in each year across all partners, wv_t^k . We do not consider *bilateral* weight-to-value ratios as Hillberry and Hummels (2000) show that bilateral weight-to-value significantly falls with distance. This suggests that the commodity composition of trade is sensitive to bilateral costs and that weight-to-value is endogenous.¹⁷ Overall, since the freight component of trade costs is higher for bulky, high weight-to-value raw materials than for manufactures, weight-to-value should be associated with higher values for θ_{ij}^k . As expected, higher transportation costs as proxied by $cfob_t^k$ and wv_t^k decrease trade integration.

Policy Variables In Column (2) we consider the role of several policy-related factors in affecting trade integration. Across countries, we include a dummy variable for Finland and Austria, FI, AT_{ij} , as these two countries were the last in our sample to join the EU (in 1995), and they appear less integrated compared to the other countries. We also explore the effect of not adopting the Euro by including a dummy variable, $noEURO_{ij,t}$, which is equal to one for Denmark and the United Kingdom from 2002 onwards when the common currency was introduced.¹⁸ As expected, non-adoption of the Euro has lowered the extent of trade integration for these two countries since 2002.

We also add a variable to capture the effects of the Schengen Agreement which abolishes physical border controls among participating EU countries. The date of the first implementation of the agreement differs across countries. Ireland and the United Kingdom have not started implementation yet. The coefficient on this variable, $Schengen_{ij,t}$, is negative and highly significant in Column (2), suggesting that the abolition of border controls among the participating countries has helped to foster trade integration, most probably through the elimination of time delays and administrative burdens that were previously experienced at borders.

We also consider policy variables that vary across industries. First, we address the role of Technical Barriers to Trade (TBTs). TBTs result from norms (regulations and standards) that affect the sale of goods in some markets by requiring specific product characteristics or production processes. Baldwin (2000) stresses the importance of TBTs in shaping trade flows between countries and industries. He argues that in the case of Europe, such barriers have become more and more visible over time, especially since tariff barriers were completely eliminated by 1968.¹⁹ In addition, TBTs are a predominant concern

¹⁶Apart from insurance and other possible discrepancies between partners, the ratio of *c.i.f.* and *f.o.b.* for intra-EU trade values mostly captures transportation costs by *road*. In 1998, 57.8 percent of total intra-EU15 trade went by road, 22.8 percent by sea, 3.9 percent by rail, 3.9 percent by air, 0.9 percent by inland waterways and 0.8 percent by pipeline (European Commission, 2000).

¹⁷Harrigan (2005) and Baldwin and Harrigan (2007) also find that export unit values are positively related to distance. ¹⁸The Euro exchange rates were fixed in 1999 but Euro notes and coins and thus a greater degree of transparency only followed in 2002. The huge literature on the trade-creating effects of currency unions raises the issue of endogeneity of the common currency dummy (Rose, 2000, Baldwin, 2006). In this paper we do not attempt to address this issue.

¹⁹As explained by Baldwin (2000, p.255), "Europe's first liberalization efforts focused on the 'easy' barriers, tariffs and quotas. With these eliminated by 1968, liberalization attention turned to TBTs."

in today's global trade negotiations and for the WTO in particular. Data on TBTs are hard to find, however, so our approach to measuring TBTs uses two different sources of information: a cross-country survey of EU managers who reveal whether they consider TBTs a problem for trade, and a ranking of industries according to the relevance of TBTs. The interacted variable, TBT_{ij}^k , increases with the extent of TBTs across countries and industries. The data appendix provides details. In Column (2) TBTs are found to lower trade integration, suggesting that some room is left for policy action and that the removal of such barriers might promote trade integration. This result is consistent with Chen (2004) who finds that TBTs in Europe are associated with larger border effects in international trade.

As is often the case, results based on survey data should be interpreted with caution. In particular, the cross-country measurement of TBTs might be problematic. For instance, if a manager states that TBTs are not a problem, this does not necessarily mean that TBTs are absent or unimportant. Instead, if that manager's company or industry is protected by TBTs initiated by his country, he might actually prefer them. This possibility might bias our cross-country measurement of TBTs.²⁰ In Column (3) we therefore report estimates when only including our industry-specific variable on TBTs, TBT^k . The results remain consistent, i.e., when TBTs are high, trade integration is low.²¹

Furthermore, it is well-established that national governments often favor domestic over foreign firms for some of their purchases, even if foreign suppliers could actually offer them a better deal (Davies and Lyons, 1996). Firms in such "public procurement" markets are hence protected from foreign imports, sometimes to such an extent that trade may be completely suppressed. In the last few years, competition has been steadily increasing in those markets, with the proportion of EU15 public procurement contracts openly advertised in the *Official Journal* of the EU steadily increasing from 8.4 percent in 1995 to 16.8 percent in 2005 (Eurostat). To investigate whether this opening of the markets has helped to lower trade frictions, we rely on time-varying cross-country data on the share of public procurement contracts advertised in the *Official Journal* and compute the average across partners. As similar data across industries are not always available, we compute an indicator variable for high public procurement sectors.²² We interact this indicator variable with the time-varying data for each country pair, denoting the variable so obtained by $Proc_{ij,t}^k$. Column (2) reveals that the opening of public procurement markets to foreign competition has indeed been successful in fostering trade integration across countries and industries and over time.

Next, Value-Added Taxes (VATs), which differ extensively across EU countries and goods, are characterized by the so-called "destination principle": VAT for a good is paid in the country where it is sold, not where it is manufactured, implying that VATs uniformly affect domestic trade and imports. In our model, the imposition of VATs by country *i* thus increases both domestic costs t_{ii} and importing costs t_{ji} by the same proportion, leaving bilateral trade integration θ_{ij}^k unchanged. In Column (2) we include the log of one plus the average VAT rate across partners at the sectoral level, VAT_{ij}^k , and interestingly find that higher VATs are associated with significantly lower trade frictions.

One possible interpretation relates to VAT fraud. Baldwin (2006) documents that VAT fraud is a very serious problem in the EU. Since the removal of Europe's internal borders in 1993, trade statistics

 $^{^{20}}$ If this is indeed the case, we will underestimate the impact of TBTs on trade integration so that our coefficients can be considered lower-bound estimates.

²¹The use of this sectoral variable on TBTs might be criticized on the grounds that it captures changes in TBTs rather than levels (see the data appendix). As in Chen (2004), we experimented using a dummy variable only, which is equal to one when TBTs are present. The results remain unaffected.

²²Examples of high public procurement industries are Shipbuilding, Rail stock, Pharmaceuticals, Aerospace, Telecoms.

are now collected by VAT authorities, creating a direct correlation between trade statistics and VAT fraud: EU firms have an incentive to over-report their exports (to get the VAT rebate) and to underreport their imports (which are subject to VAT). This inflates export statistics relative to import statistics. As a result, in the countries and industries where VATs are high, the value of θ_{ij}^k might be lower simply because intra-EU export statistics are inflated due to VAT fraud.²³

Other Costs The new trade literature on heterogeneous firms rationalizes why some firms export and others do not. In particular, the models by Melitz (2003), Melitz and Ottaviano (2008) and others show that only the most productive firms participate in foreign markets. Only those firms are productive enough to cover the fixed costs of exporting or to withstand tough competition on foreign markets. Due to this selection effect we should observe that ceteris paribus, more productive industries trade more on foreign markets than less productive industries, i.e., we should observe a negative relationship between productivity and trade frictions θ_{ij}^k . In Column (4) of Table 3 we therefore include average real labor productivity across industries and countries. Moreover, a growing empirical literature shows that as trade costs fall, less efficient firms exit from the market and average industry productivity increases (for example, see Bernard, Jensen and Schott, 2006, for the U.S. and Greenaway and Kneller, 2008, for the UK). To mitigate potential endogeneity problems we lag productivity by one year, $Prod_{ij,t-1}^k$, reducing the number of observations by one cross section. The negative relationship between productivity and θ_{ij}^k confirms the prediction that more productive industries trade more on foreign markets.²⁴

We also include a dummy variable in Column (4) that controls for the five zero trade observations in our sample, $Zeros_{ij,t}^k$. We find that trade frictions between two countries in a given industry are stronger whenever either of them does not export. As models of heterogeneous firms typically predict that firms only export if they are able to cover the fixed costs of exporting, the result might reflect the role of these fixed costs.

Controls Finally, we control for some measurement issues and the possibility that the magnitude of θ_{ij}^k may be affected by the nature of trade. In particular, the value of θ_{ij}^k computed under intra-industry trade will be lower than under comparative advantage driven by technology or factor endowment differences across countries, even if the actual friction is the same.²⁵ To control for this potential bias we compute the (absolute) difference in capital shares between countries and industries in order to proxy for differences in factor endowments. For a larger differential in capital shares, trade is more likely based on comparative advantage, leading to an overestimated θ_{ij}^k . This variable, $KS_{ij,t}^k$, displays the expected positive (but insignificant) coefficient in Column (5) but its inclusion leaves the coefficients on the other variables virtually unchanged.

The previous reasoning presumes that differences in factor endowments can be fully captured by differences in capital shares, the latter being computed at the *industry* level. One obvious limitation stems from the use of data disaggregated at the level of industries and not at the level of products.

²³ If the intra-EU exports of country *i* are inflated because of VAT fraud, then x_{ij}^k will be high relative to x_{ii}^k , leading to a lower value of θ_{ij}^k .

²⁴We also tried to use data on productivity for a country that is not included in our sample, i.e., the US. But those data are aggregated at the 2-digit level and proved insignificant in explaining trade integration across EU countries, possibly due to the higher level of aggregation.

 $^{^{25}}$ With intra-industry trade, the denominator of (14) will be given by the product of two balanced trade flows. But with comparative advantage, the denominator of (14) will be given by the product of two unbalanced trade flows, leading to a relatively small magnitude in the denominator and thus overestimated trade frictions.

Industry classifications inevitably aggregate many different types of products into one single category so that the volume of intra-industry trade can appear more important than it actually is. For instance, many different types of steel are produced, from flat-rolled to specialty steels, and it may be that the production of some types of steel requires resources or technologies in which one country has a comparative advantage. However, since all these types of steel are aggregated into one category, it appears as if the countries export and import almost identical products while in reality they export one type of steel and import another type.

Thus, the larger the number of varieties in each industry category, the more likely the industry aggregates trade flows with different comparative advantages in different varieties, but in the data this will show up as intra-industry trade, i.e., more balanced trade flows between the two countries. The resulting θ_{ij}^k is therefore expected to be smaller because the denominator in equation (14) is the product of two more or less balanced (intra-industry) trade flows at the *industry* level, masking imbalances of (inter-industry) trade flows at the *product* level.²⁶ To control for this aggregation bias, we include the number of product categories within each industry, $Goods^k$. In Column (5) it is indeed the case that a larger number of varieties is associated with a lower value for θ_{ij}^k . But the inclusion of this control hardly affects other estimates.

In Column (6) we replicate the specification of Column (5) but replace the country-varying variable on TBTs by the sectoral variable, TBT^k . It is reassuring that all results remain virtually unchanged, providing some evidence that they are not biased by the use of cross-country survey data on TBTs.

Finally, Table 4 reports a variance decomposition to illustrate the quantitative contribution of each factor in explaining the variation of θ_{ij}^k . Following Fields (2003), the contribution of each explanatory variable x_m to the total variance of $\theta_{ij,t}^k$ is calculated as $c_m = \beta_m \cos\left(x_m, \theta_{ij,t}^k\right) / var\left(\theta_{ij,t}^k\right)$ where β_m is the partial regression coefficient of $\theta_{ij,t}^k$ on the explanatory variable x_m (holding all other explanatory variables constant). Columns (1) and (2) correspond to regressions (5) and (6) in Table 3, respectively.

Column (1) shows that regression (5) in Table 3 explains 80.8 percent of the variation in $\theta_{ij,t}^k$. 42.8 percent is attributable to the 3-digit industry fixed effects and 0.01 percent to the year fixed effects. Of the 38 percent that the remaining regressors explain, geography and transport costs alone are responsible for about 25 percent of the variation in θ_{ij}^k , the most important factor being weight-tovalue (16.7 percent), followed by bilateral distance (5.4 percent). Policy variables explain 7.6 percent of the variation in θ_{ij}^k , which is far from negligible. TBTs are the most important factor (4.5 percent), while public procurement, Schengen and opting out of the Euro only play minor roles. The number of product categories within each industry explains up to 5.4 percent of the variation in θ_{ij}^k . Finally, the decomposition reported in Column (2), which corresponds to regression (6) in Table 3, gives similar results.

4.2 How Our Approach Differs from Standard Gravity

We now compare our specification to the standard gravity approach and illustrate the empirical pitfalls of not taking the theory seriously. We believe this is an important and necessary exercise.

Although specification (16) is related to standard gravity estimation, there are three important differences. First, standard gravity uses output $y_i^k y_j^k$ as the size variable. In contrast, the trade

²⁶ The product of two balanced trade flows yields a relatively large magnitude in the denominator and thus a relatively low θ_{ij}^k .

integration measure θ_{ij}^k includes domestic trade $x_{ii}^k x_{jj}^k$ because this nets out the multilateral resistance variables. Second, correctly specified gravity equations have to include time-varying industry fixed effects for each country pair to control for multilateral resistance variables (Baldwin and Taglioni, 2006). But a practical problem is that these fixed effects are perfectly collinear with many explanatory variables of interest such as $cfob_t^k$ and wv_t^k in Table 3. We avoid this problem since θ_{ij}^k nets out the time-varying multilateral resistance variables. Third, our theory calls for trade cost elasticities that vary across industries. Those are embodied in θ_{ij}^k .

We now evaluate the bias that arises in standard gravity equations due to the omission of multilateral resistance and the assumption of coefficient homogeneity across industries. In its simplest form (henceforth the "naive" model), the traditional gravity model relates bilateral trade to output and bilateral trade costs where the trade cost elasticity α_2 is assumed to be the same across industries:

$$\ln\left(x_{ij}^{k}x_{ji}^{k}\right) = \alpha_{1} \ln\left(y_{i}^{k}y_{j}^{k}\right) + \alpha_{2} \ln\left(t_{ij}^{k}t_{ji}^{k}\right) + \epsilon_{ij}^{k}$$

$$\tag{17}$$

We drop time subscripts for simplicity. Compare the naive specification in equation (17) to the loglinearized version of micro-founded gravity equation (12):

$$\ln\left(x_{ij}^{k}x_{ji}^{k}\right) = \beta_{1} \ln\left(x_{ii}^{k}x_{jj}^{k}\right) + \beta_{2}\left(1-\sigma_{k}\right)\ln\left(t_{ij}^{k}t_{ji}^{k}\right) - \beta_{3}\left(1-\sigma_{k}\right)\ln\left(t_{ii}^{k}t_{jj}^{k}\right) + \epsilon_{ij}^{k} \tag{18}$$

The size variable is domestic trade instead of output. Moreover, equation (18) includes domestic trade costs and the trade cost coefficients now vary across industries.²⁷ An alternative micro-founded gravity specification can be obtained by log-linearizing equation (11):

$$\ln\left(x_{ij}^k x_{ji}^k\right) = \ln\left(y_i^k y_j^k\right) + \ln\left(\frac{x_i^k x_j^k}{y^k y^k}\right) + (1 - \sigma_k)\ln\left(t_{ij}^k t_{ji}^k\right) + (\sigma_k - 1)\left(\Pi_i^k P_i^k \Pi_j^k P_j^k\right)$$

To control for the unobservable multilateral resistance variables $\prod_i^k P_i^k \prod_j^k P_j^k$, we include fixed effects for each industry (at the 4-digit level) in each year and separately for countries *i* and *j*, denoted by γ_i^k and γ_j^k . Such fixed effects also wipe out the domestic demand terms (if included separately for each country) and world output in industry *k*. To ease the comparison with the naive specification, we keep the domestic output terms so that we estimate

$$\ln\left(x_{ij}^{k}x_{ji}^{k}\right) = \delta_{1} \ln\left(y_{i}^{k}y_{j}^{k}\right) + \delta_{2}\left(1 - \sigma_{k}\right)\ln\left(t_{ij}^{k}t_{ji}^{k}\right) + \gamma_{i}^{k} + \gamma_{j}^{k} + \epsilon_{ij}^{k}$$
(19)

To keep our exposition as simple as possible, we assume that trade costs (domestic and international) are proxied by the log of distance only. We then estimate the naive specification (17) and compare its distance elasticity to the ones estimated in the micro-founded specifications (18) and (19). Table 5 reports our results. The upper panel refers to naive gravity while the two lower panels refer to the micro-founded models. For all panels, Column (1) assumes homogeneity on the trade cost coefficient across industries (bilateral and domestic). We relax this assumption in Column (2) where we report the average of the elasticities estimated for individual industries. In Columns (3) and (4), we replicate the specifications in Columns (1) and (2) but we allow the output variables, domestic distances and domestic trade to enter separately for each country.

²⁷Combes, Lafourcade and Mayer (2005, equation 13) estimate a form of equation (18) where they constrain β_1 to unity. But they use aggregate trade data so that by construction the trade cost elasticity cannot vary across industries.

As can be seen in Column (1) of the upper panel, the naive gravity equation yields an absolute distance elasticity of 3.589, which increases slightly to 3.741 once we allow for coefficient heterogeneity in Column (2).²⁸ The sectoral output regressor displays a coefficient slightly larger than the unity value predicted by theory. All elasticities remain similar in Columns (3) and (4).

The middle panel turns to the micro-founded specification in (18). The coefficients on domestic distances and domestic trade flows are significant with the expected signs. But the coefficients on bilateral distance are markedly different. In Column (1), the absolute magnitude of the distance elasticity is large and equal to 4.514. It further increases to 5.131 once we allow for coefficient heterogeneity, as required by theory.²⁹ As highlighted by Anderson and van Wincoop (2003) and Baldwin and Taglioni (2006), the omission of multilateral resistance biases the magnitude of trade cost variables. The magnitude of the distance elasticity in the naive specification is underestimated by 30 percent (3.589/5.131-1 = -0.30) and by 28 percent if we allow the other controls to enter separately for each country (3.629/5.045-1 =-0.28). The omitted variable bias goes in the direction predicted by theory.³⁰

The lower panel focuses on the micro-founded specification with fixed effects. The fixed effects do a good job in eliminating the bias on the distance coefficient. The distance elasticity is 4.204 in Column (1) and increases to 4.653 when we allow for coefficient heterogeneity in Column (2). In absolute terms, those elasticities are not as large as in the other micro-founded specification but the difference is relatively small (between 7 and 9 percent).

In summary, we show that omitting multilateral resistance (and assuming coefficient homogeneity across industries) yields biased distance elasticities, and the bias can be large (up to 30 percent in our sample). Our original specification (16) is consistent with the estimation of the micro-founded gravity equation (18), which differs from standard gravity regressions by including domestic trade as the size variable. In principle, fixed effect estimation in equation (19) is a viable micro-founded alternative but it obviates the identification of many explanatory variables of interest such as $cfob_t^k$ and wv_t^k because they are linear combinations of the fixed effects.

4.3 The Evolution of Trade Integration over Time

To investigate the evolution of trade integration over time, we run the following regression

$$\ln \theta_{ij,t}^k = \gamma_{ij}^k + \psi_t + \alpha \, \ln w v_{ij,t}^k + \varepsilon_{ij,t}^k \tag{20}$$

where γ_{ij}^k are country pair dummies interacted with industry intercepts (at the 4-digit level) to control for systematic differences between trade partners in each industry. The *bilateral* weight-to-value $wv_{ij,t}^k$ controls for compositional change over time (see Hummels, 2007).³¹ The evolution of θ_{ij}^k is given by the estimated coefficients on the year dummies ψ_t . The annualized growth rate of these coefficients

²⁸The magnitude of this coefficient is larger than that typically obtained in the literature because the dependent variable is the product of bilateral exports and not unidirectional bilateral exports.

²⁹ Chen (2004) also finds that the inclusion of fixed effects to control for multilateral resistance leads to larger magnitudes of the distance coefficients.

³⁰The naive gravity equation yields a distance elasticity $\widehat{\alpha_2}$ such that $E(\widehat{\alpha_2}) = \alpha_2 + ab$ where $\alpha_2 < 0$ is the true elasticity, $a = \partial \ln (\prod_i P_i \prod_j P_j) / \partial \ln (t_{ij} t_{ji}) > 0$ and $b = (\sigma_k - 1) > 0$. The bias, equal to ab > 0, therefore leads to an underestimation of the absolute magnitude of the distance elasticity.

³¹In contrast to Section 4.1, we now consider the *bilateral* weight-to-value as distance is not included in the regression.

obtained from estimating equation (20) is equal to -0.94 percent over our pooled sample with a *t*-statistic of $11.48.^{32}$ This result is highly significant and we conclude that trade integration has on average improved between 1999 and 2003.

We also estimate equation (20) separately for each industry. Figure 2 plots the annualized growth rates for the individual industries, distinguishing between significant (at the ten percent level) and insignificant growth rates.³³ About half of the industries did not experience any significant change over time (82 industries out of 160). However, among the industries that did, the vast majority (69 out of 78 significant) experienced an increase in trade integration. Games and toys display the strongest improvement, while trade integration worsened for Articles of concrete followed by Paper and Explosives. Note that according to equation (14), in the few cases where θ_{ij}^k increases over time, the worsening in bilateral trade integration could reflect a fall in domestic trade costs, $t_{ii}^k t_{jj}^k$, rather than an increase in bilateral trade costs, $t_{ij}^k t_{ij}^k$.

Lacking any product or firm-level data, we are not able to investigate whether the deepening of trade integration over time is due to the extensive or intensive margin of trade. On the one hand, Hummels and Klenow (2005) report evidence in favor of the extensive margin, while Helpman, Melitz and Rubinstein (2008) and Besedes and Prusa (2007) argue that the intensive margin is quantitatively more important. To investigate this issue we would ideally need to observe more years and more highly disaggregated data. We leave this question for future research.

4.4 Explaining the Growth in Trade

Baier and Bergstrand (2001) try to explain the growth of world trade by tracing two fundamental channels through which trade can grow -(1) the growth of income and (2) reductions in trade barriers. For that purpose, they estimate a gravity equation in first differences and use the gravity coefficients to extrapolate the contributions of income growth and trade barrier reductions to the average growth of trade in their sample.

We use our model to perform a similar exercise of decomposing the growth of trade into these two contributions. But our approach differs on three counts. First, our approach is considerably easier to implement because our solution for time-varying multilateral resistance variables allows us to eliminate the unobservable and highly nonlinear price indices that appear in Baier and Bergstrand's (2001) gravity equation. Second, we use industry-level trade data instead of aggregate trade data. Third, our decomposition is not restricted to the sample average because we are able to provide decompositions for specific bilateral observations in each industry. These finer decompositions allow us to uncover a substantial degree of heterogeneity across various cuts of our dataset.

Our decomposition is based on gravity equation (11), which can be rearranged as

$$\begin{aligned}
x_{ij}^{k} x_{ji}^{k} &= y_{i}^{k} y_{j}^{k} \left(\frac{t_{ij}^{k} t_{ji}^{k}}{t_{ii}^{k} t_{jj}^{k}} \right)^{1-\sigma_{k}} \frac{x_{i}^{k} x_{j}^{k}}{y^{k} y^{k}} \left(\frac{t_{ii}^{k} t_{jj}^{k}}{\Pi_{i}^{k} P_{i}^{k} \Pi_{j}^{k} P_{j}^{k}} \right)^{1-\sigma_{k}} \\
&= y_{i}^{k} y_{j}^{k} \left(\theta_{ij}^{k} \right)^{2(1-\sigma_{k})} \frac{x_{i}^{k} x_{j}^{k}}{y^{k} y^{k}} \left(\frac{t_{ii}^{k} t_{jj}^{k}}{\Pi_{i}^{k} P_{i}^{k} \Pi_{j}^{k} P_{j}^{k}} \right)^{1-\sigma_{k}}
\end{aligned}$$
(21)

 $^{^{32}\}mathrm{The}$ significance is calculated using the delta method.

 $^{^{33}}$ We can only estimate equation (20) separately for 160 industries because of too few degrees of freedom for the other 6 industries. To conserve space we do not report the annualized growth rates for each industry but they are available upon request.

Using the solution for the multilateral resistance terms in (10), we can express the last two fractions on the right-hand side of equation (21) in terms of observable variables, yielding

$$x_{ij}^{k} x_{ji}^{k} = y_{i}^{k} y_{j}^{k} \left(\theta_{ij}^{k}\right)^{2(1-\sigma_{k})} \frac{x_{ii}^{k}}{y_{i}^{k}} \frac{x_{jj}^{k}}{y_{j}^{k}}$$
(22)

We now take logs and first differences of equation (22) and divide by the left-hand side to obtain our decomposition equation

$$100\% = \underbrace{\frac{\Delta \ln \left(y_i^k y_j^k\right)}{\Delta \ln \left(x_{ij}^k x_{ji}^k\right)}}_{\text{(First factor)}} + \underbrace{\frac{2(1 - \sigma_k)\Delta \ln \left(\theta_{ij}^k\right) + \Delta \ln \left(\frac{x_{ii}^k}{y_i^k} \frac{x_{jj}^k}{y_j^k}\right)}{\Delta \ln \left(x_{ij}^k x_{ji}^k\right)}}_{\text{(Second factor)}}$$
(23)

The denominator on the right-hand side of equation (23), $\Delta \ln \left(x_{ij}^k x_{ji}^k\right)$, is the growth of bilateral trade between countries *i* and *j* in industry *k*. The first factor thus represents the contribution of output growth, $\Delta \ln \left(y_i^k y_j^k\right)$, to the growth of bilateral trade.

The second factor is the contribution of trade integration in that industry. If bilateral trade frictions decline relative to domestic trade frictions, then $\Delta \ln \left(\theta_{ij}^k\right) < 0$. As $\sigma_k > 1$, it follows that the contribution of declining bilateral trade frictions is positive (given positive trade growth $\Delta \ln \left(x_{ij}^k x_{ji}^k\right) > 0$). The change in the domestic trade to output ratios, $\Delta \ln \left(\frac{x_{ii}^k x_{jj}^k}{y_i^k}\right)$, reflects changes in multilateral factors. In particular, from equation (21) it can be seen that this captures changes in world demand shares and changes in multilateral resistance relative to domestic trade costs.

Based on annual changes, we first calculate the two factors in equation (23) for each country pairindustry observation. To avoid giving undue weight to quantitatively unimportant observations that only represent a small fraction of total trade, we then compute a weighted average where the weights are the products of sectoral bilateral exports in the initial year 1999.³⁴

Table 6 reports the average growth in trade and the contributions of the two factors. In the full sample, trade grew on average by 7.96 percent over the period.³⁵ 42 percent of this trade expansion can be accounted for by changes in the trading partners' output growth, whereas changes in trade integration account for 58 percent of the growth in trade. These proportions are different from those reported by Baier and Bergstrand (2001) who argue that two-thirds of the growth in trade amongst OECD countries between 1958 and 1988 can be explained by the growth of output. Our results therefore suggest that trade integration amongst EU countries over our sample period has progressed relatively faster. Likewise, Jacks, Meissner and Novy (2008) find that more than half of the global trade boom between 1870 and 1913 is attributable to reductions in trade costs.

The next rows of the table focus on various subsamples.³⁶ These finer decompositions show that

 $^{^{34}}$ Unweighted decompositions are very similar to the weighted decompositions reported in the paper and are available upon request.

³⁵To be precise, this is the average annual growth rate of the product of bilateral trade, $x_{ij}^k x_{ji}^k$, weighted as described above.

³⁶We do not report decompositions for VAT, capital shares or the number of goods in each sector as those are only included in our earlier regressions to control for possible measurement error.

the extent to which trade expansion is affected by trade integration is strongly limited by geography and transportation costs, both across countries and industries. The contribution of trade integration is larger for countries that are separated by short distances (79 percent) or adjacent to each other (72 percent), and for goods that are easier and cheaper to transport, i.e., with a low *c.i.f./f.o.b.* value (74 percent) and a low weight-to-value ratio (63 percent).³⁷ The latter two findings are consistent with some recent evidence showing that the relative price of air to non-air shipping has declined over time (Hummels, 2007, Choate, 2008). Such a decrease in relative costs should encourage trade in air transported goods, which are indeed characterized by a low weight-to-value (or low *c.i.f./f.o.b.*). Finally, Table 6 shows that speaking a common language, which is of course correlated with short distances and adjacency, is also associated with a higher contribution of trade integration (76 percent).

The effectiveness of trade integration in promoting the growth in trade is also influenced by policy factors. On average, the contribution of trade integration is slightly stronger for Finland and Austria than for the other countries in the sample (64 percent against 56 percent).³⁸ This indicates that Finland and Austria, who only joined the EU in 1995, experienced a somewhat faster rate of integration. Interestingly, non-Eurozone countries only experienced a very small increase (equal to 0.32 percent) in trade with Eurozone countries since 2002, which is consistent with our earlier finding that the introduction of the Euro has hampered the integration of those countries. Moreover, changes in trade integration only had a minor effect on the trade growth of Euro outsiders (15 percent versus 79 percent for Eurozone countries). This indicates that in contrast to the Eurozone, the trade growth of Euro outsiders was predominantly driven by secular output growth, not by improvements in trade integration.³⁹

We then repeat the exercise for subsamples in which none, one or both of the partners have implemented the Schengen Agreement. Consistent with our previous findings, the elimination of physical borders seems to have greatly fostered trade. Trade integration accounts for 76 percent of the growth in trade when both partners have implemented the Agreement but for only 21 percent when just one has joined. Trade integration has actually declined for the pairs in which none of the partners has implemented the Agreement so that the contribution is negative (-44 percent).

The pattern of the results is somewhat different once we focus on industry-specific characteristics such as TBTs and the transparency of public procurement. In all three cases, changes in trade integration have been most effective where trade barriers were initially higher, i.e., in industries affected by high TBTs (83 percent) and markets with a low transparency in public procurement (76 percent). This indicates that the process of EU integration has been to some degree successful in overcoming such barriers. Finally, trade expansion in highly productive industries is predominantly driven by changes in trade integration (88 percent), whereas changes in output are still the main driver of trade growth in low productivity industries.

5 Robustness

In this section we report a number of alternative specifications that we implement to ensure the robustness of our conclusions. First, we estimate our main specification by Poisson Pseudo-Maximum Likelihood (PPML) as recommended by Santos Silva and Tenreyro (2006). They show that under

 $^{^{37}}$ The sample is split at the mean value of the variable in question in 1999, denoted by † in Table 6.

³⁸Given that those relative contributions are calculated and not estimated, we are not able to test whether the difference between these two numbers is significant.

³⁹As we only focus on the post-2002 period, note that the sample size is much smaller than in the full sample.

heteroskedasticity, coefficient estimates from log-linear regressions estimated by OLS can be strongly biased. Martin and Pham (2008) confirm that the PPML estimator solves the heteroskedasticity bias problem, but instead show that this estimator yields severely biased estimates when the dependent variable takes on many zero values. As our dependent variable has virtually no zero values but heteroskedasticity might be present, the PPML estimator appears adequate as a robustness check. But as Column (1) of Table A1 shows, there are no qualitative changes and generally only small quantitative changes in the estimation results compared to those obtained with OLS.

Second, we verify in Columns (2) and (3) that the results are robust to the inclusion of country pair fixed effects interacted with year dummies and to the exclusion of the series that contain zero trade observations.⁴⁰ We also want to make sure that our results are not driven by observations that are economically small. For instance, in Table 3 we find that the effect of remaining outside the Eurozone is significant. But the dummy that captures the effect is equal to unity in only 1,796 cases out of a total of 15,040 observations (i.e., 12 percent of the sample), and roughly half of those cases involve Denmark, one of the smallest economies in our sample. We therefore run a regression where the observations are weighted by the product of sectoral bilateral exports in the initial year 1999. The results, reported in Column (4), reveal that a significant Eurozone effect is still present.

With panel data, one issue relates to the computation of θ_{ij}^k in *real* terms. Ideally, for deflating we would need domestic price indices for the numerator and export price indices for the denominator in equation (14). Export price indices are not available, so common practice it to use domestic deflators instead (Baldwin and Taglioni, 2006). In that case, the deflators in (14) cancel out and using nominal or real variables yields the same values for θ_{ij}^k (which is the approach we have followed in the paper). Since there are reasons to believe that domestic and export price indices differ from each other, for instance in the case of pricing-to-market practices, we need to tackle this empirical problem up-front.

One way of doing this is to include a set of time-varying sector (3-digit) fixed effects, which leaves our results unaffected (Column 5). A second way is to run cross-sectional regressions for each year, as we do in Table A2. This approach comes at the cost of strongly reducing the number of observations for each regression so that not all variables are significant in all years, but overall the results largely hold up.⁴¹ It is interesting that the magnitude of the coefficient on the non-Eurozone dummy substantially increases between 2002 and 2003, suggesting that the negative effect of not adopting the Euro might have strengthened over time. There is also some evidence that the importance of TBTs increased in the last three years. Finally, the coefficients on distance remain stable.⁴²

In order to compare how θ_{ij}^k performs relative to alternative measures of trade integration, we use the log inverse *phi*-ness of trade as a dependent variable in Column (6) of Table A1. This implicitly assumes that the elasticity of substitution is equal to 1.5 for all industries so that the exponent in equation (14) becomes unity. Assuming a homogeneous elasticity of substitution across industries reduces the adjusted R^2 considerably by about twelve percentage points to 0.693. Moreover, although

 $^{^{40}}$ In the case of zero trade observations, we drop the entire series to keep a balanced sample over time. This results in 16 fewer observations in Column (3) as the five zero trade observations correspond to four different series only.

⁴¹Lagged productivity is omitted for 1999.

 $^{^{42}}$ The literature that examines the evolution of distance effects over time yields mixed results. Coe, Subramanian and Tamirisa (2007) find declining distance effects once they estimate a non-linear version of the gravity equation with an additive error term. In contrast, Berthelon and Freund (2008) find an increasing impact of distance as the effect of distance remains unchanged in 75 percent of their industries but increases in the remaining industries. For an overview, see Disdier and Head (2008).

their signs remain mostly unchanged, the size of most estimated coefficients is radically different. In addition, not being a member of the Eurozone, TBTs and sharing a common border are no longer significantly associated with trade integration.⁴³

6 Concluding Remarks

This paper explores the sources and size of trade impediments. Measuring these impediments directly is often difficult because of data limitations. Instead, we indirectly infer trade impediments from observable trade flows. For this purpose we use the gravity framework pioneered by Anderson and van Wincoop (2003) and apply it to industries with heterogeneous trade costs and heterogeneous elasticities of substitution. The model yields a micro-founded measure of bilateral trade integration that controls for time-varying multilateral resistance and that can be applied to disaggregated panel data. We show that this trade integration measure is also consistent with the Ricardian model by Eaton and Kortum (2002) and recent heterogeneous firms models.

From an empirical perspective, we use this micro-founded measure to explore the key determinants of trade integration across countries and industries in the European Union. We show that cross-country trade integration is lower for those countries in our sample that joined the EU most recently and that opted out of the Euro common currency, but it is higher for countries that have implemented the Schengen Agreement. We also document a considerable variation in trade integration across industries. Consistent with the literature on firm heterogeneity, we find that industries with higher productivity trade more on international markets. Trade integration is hampered by transportation costs as captured by industry-specific weight-to-value and c.i.f./f.o.b. ratios. Trade integration is also severely hampered by policy factors, in particular Technical Barriers to Trade and intransparent public procurement procedures. Our findings suggest for public policy that gains from improved international trade integration are possible through the elimination of TBTs and through more transparent advertising of public procurement contracts.

Moreover, we find that trade integration has improved on average between 1999 and 2003. We find that roughly 58 percent of the growth in trade can be accounted for by improvements in trade integration through a decrease in bilateral and multilateral trade barriers, while the rest can be attributed to increases in manufacturing output.

To conclude, our measure yields sensible systematic differences in trade integration across industries, suggesting that modeling trade costs as a "one-fits-all" impediment is clearly at odds with empirical evidence. Instead, when dealing with industry-level data it is important to allow for trade cost heterogeneity across industries.

Ultimately, of course even highly disaggregated industry-level data "can never be as fine as reality, so some degree of aggregation bias is inevitable" (Anderson and van Wincoop, 2004, p.729). Thus, the use of industry-level data precludes us from exploring the determinants and change in trade impediments at the firm or product level. In our view, analyzing trade barriers at these very fine levels of disaggregation is a natural and important next step.

⁴³We also tried using the elasticities from Broda and Weinstein (2006) for the US in computing θ_{ij}^k . However, this led to large outliers in θ_{ij}^k and the overall results were weaker. The most likely reason is that those elasticities are much more finely disaggregated and the classifications are thus harder to match. The results are available upon request.

References

Anderson J.E. and E. van Wincoop, 2002, Borders, Trade, and Welfare, *Brookings Trade Forum*. In: Collins and Rodrik, Eds., Washington: Brookings Institution, 207-244.

Anderson J.E. and E. van Wincoop, 2003, Gravity with Gravitas: A Solution to the Border Puzzle, *American Economic Review* 93(1), 170-192.

Anderson J.E. and E. van Wincoop, 2004, Trade Costs, Journal of Economic Literature 42(3), 691-751.

Arkolakis K., Demidova S., Klenow P.J. and A. Rodríguez-Clare, 2008, Endogenous Variety and the Gains from Trade, *American Economic Review* 98(2), Papers & Proceedings, 444-450.

Baier S.L. and J.H. Bergstrand, 2001, The Growth of World Trade: Tariffs, Transport Costs, and Income Similarity, *Journal of International Economics* 53(1), 1-27.

Baldwin R.E., 2000, Regulatory Protectionism, Developing Nations, and a Two-Tier World Trade System, *Brookings Trade Forum*. In: Collins and Rodrik, Eds., Washington: Brookings Institution, 237-293.

Baldwin R.E., 2006, The Euro's Trade Effects, European Central Bank Working Paper 594.

Baldwin R.E., Forslid R., Martin P., Ottaviano G.I.P. and F. Robert-Nicoud, 2003, The Core-Periphery Model: Key Features and Effects. In: Heijdra and Brakman, Eds., The Monopolistic Competition Revolution in Retrospect, Cambridge University Press, 213-235.

Baldwin R.E. and J. Harrigan, 2007, Zeros, Quality and Space: Trade Theory and Trade Evidence, *NBER Working Paper* 13214.

Baldwin R.E. and D. Taglioni, 2006, Gravity for Dummies and Dummies for Gravity Equations, *NBER* Working Paper 12516.

Bergstrand J., 1989, The Generalized Gravity Equation, Monopolistic Competition, and the Factor-Proportions Theory in International Trade, *Review of Economics and Statistics* 71(1), 143-153.

Bergstrand J., 1990, The Heckscher-Ohlin-Samuelson Model, the Linder Hypothesis and the Determinants of Bilateral Intra-Industry Trade, *Economic Journal* 100(403), 1216-1229.

Bernard A.B., Jensen J.B. and P.K. Schott, 2006, Trade Costs, Firms and Productivity, *Journal of Monetary Economics* 53(5), 917-937.

Berthelon M. and C. Freund, 2008, On the Conservation of Distance in International Trade, *Journal of International Economics* 75(2), 310-320.

Besedeš T. and T.J. Prusa, 2007, The Role of Extensive and Intensive Margins and Export Growth, *NBER Working Paper* 13628.

Broda C. and D.E. Weinstein, 2006, Globalization and the Gains from Variety, *Quarterly Journal of Economics* 121(2), 541-585.

Chaney T., 2008, Distorted Gravity: The Intensive and Extensive Margins of International Trade, American Economic Review 98(4), 1707-1721. Chen N.A., 2004, Intra-National Versus International Trade in the European Union: Why Do National Borders Matter?, *Journal of International Economics* 63(1), 93-118.

Choate D.A., 2008, Relative Transport Price Movements and U.S. Imports, Columbia University, job market paper.

Coe D.T., Subramanian A. and N.T. Tamirisa, 2007, The Missing Globalization Puzzle: Evidence of the Declining Importance of Distance, *IMF Staff Papers* 54(1), 34-58.

Combes P.-P. and M. Lafourcade, 2005, Transport Costs: Measures, Determinants, and Regional Policy Implications for France, *Journal of Economic Geography* 5(3), 319-349.

Combes P.-P., Lafourcade M. and T. Mayer, 2005, The Trade-Creating Effects of Business and Social Networks: Evidence from France, *Journal of International Economics* 66(1), 1-29.

Costinot A. and I. Komunjer, 2007, What Goods Do Countries Trade? New Ricardian Predictions, *NBER Working Paper* 13691.

Davies S. and B. Lyons, 1996, Industrial Organization in the European Union – Structure, Strategy, and the Competitive Mechanism, Oxford University Press.

Deardorff A.V., 1998, Determinants of Bilateral Trade: Does Gravity Work in a Neoclassical World?. In: The Regionalization of the World Economy, Frankel, Ed., Chicago: University of Chicago Press, 7-22.

Disdier A.C. and K. Head, 2008, The Puzzling Persistence of the Distance Effect on Bilateral Trade, *Review of Economics and Statistics* 90(1), 37-48.

Eaton J. and S. Kortum, 2002, Technology, Geography and Trade, *Econometrica* 70(5), 1741-1779.

European Commission, 1998, Technical Barriers to Trade. The Single Market Review, Subseries III: Dismantling of Barriers 1.

European Commission, 2000, EU Transport in Figures – Statistical Pocket Book, Directorate General for Energy and Transport.

European Commission, 2006, Internal Market: Opinions and Experiences of Businesses in EU-15, *Flash Eurobarometer 180*, TNS Sofres/EOS Gallup Europe.

European Commission, 2008, VAT Rates Applied in the Member States of the European Community, DOC/2412/2008 – EN.

Evans C.L., 2003, The Economic Significance of National Border Effects, *American Economic Review* 93(4), 1291-1312.

Evans C.L. and J. Harrigan, 2005, Distance, Time, and Specialization: Lean Retailing in General Equilibrium, *American Economic Review* 95(1), 292-313.

Evenett S.J. and W. Keller, 2002, On Theories Explaining the Success of the Gravity Equation, *Journal of Political Economy* 110(2), 281-316.

Feenstra R.C., Markusen J.R. and A.K. Rose, 2001, Using the Gravity Equation to Differentiate among Alternative Theories of Trade, *Canadian Journal of Economics* 34(2), 430-447.

Fields G.S., 2003, Accounting for Income Inequality and Its Change: A New Method, With Application to the Distribution of Earnings in the United States, *Research in Labor Economics* 22, 1-38.

Greenaway D. and R. Kneller, 2008, Exporting, Productivity and Agglomeration, *European Economic Review* 52(5), 919-939.

Grossman, G.M., 1998, Comment. In: The Regionalization of the World Economy, Frankel, Ed., Chicago: University of Chicago Press, 29-31.

Hanson G.H. and C. Xiang, 2004, The Home-Market Effect and Bilateral Trade Patterns, *American Economic Review* 94(4), 1108-1129.

Harrigan J., 1993, OECD Imports and Trade Barriers in 1983, Journal of International Economics 35(1/2), 91-111.

Harrigan J., 1996, Openness to Trade in Manufactures in the OECD, Journal of International Economics 40(1/2), 23-39.

Harrigan J., 2005, Airplanes and Comparative Advantage, NBER Working Paper 11688.

Head K. and T. Mayer, 2000, Non-Europe: The Magnitude and Causes of Market Fragmentation in the EU, *Weltwirtschaftliches Archiv* 136 (2), 285-314.

Head K. and T. Mayer, 2004, The Empirics of Agglomeration and Trade, *Handbook of Regional and Urban Economics*, Chapter 59, North-Holland.

Head K. and J. Ries, 2001, Increasing Returns Versus National Product Differentiation as an Explanation for the Pattern of US-Canada Trade, *American Economic Review* 91(4), 858-876.

Helpman E., Melitz M.J. and Y. Rubinstein, 2008, Estimating Trade Flows: Trading Partners and Trading Volumes, *Quarterly Journal of Economics* 123(2), 441-487.

Hillberry R. and D. Hummels, 2000, Explaining Home Bias in Consumption: Production Location, Commodity Composition and Magnification, Purdue University, mimeo.

Hummels D., 2001a, Toward a Geography of Trade Costs, Purdue University, mimeo.

Hummels D., 2001b, Time as a Trade Barrier, Purdue University, mimeo.

Hummels D., 2007, Transportation Costs and International Trade in the Second Era of Globalization, Journal of Economic Perspectives 21(3), 131-154.

Hummels D. and P.T. Klenow, 2005, The Variety and Quality of a Nation's Exports, *American Economic Review* 95(3), 704-723.

Jacks D.S., C.M. Meissner and D. Novy, 2008, Trade Costs, 1870-2000, American Economic Review 98(2), Papers & Proceedings, 529-534.

Limão N. and A.J. Venables, 2001, Infrastructure, Geographical Disadvantage and Transport Costs, World Bank Economic Review 15, 451-479.

Martin W. and C.S. Pham, 2008, Estimating the Gravity Model when Zero Trade Flows Are Frequent, mimeo.

McCallum J., 1995, National Borders Matter: Canada-US Regional Trade Patterns, American Economic Review 85(3), 615-623.

Melitz M.J., 2003, The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity, *Econometrica* 71(6), 1695-1725.

Melitz M.J. and G.I.P. Ottaviano, 2008, Market Size, Trade, and Productivity, *Review of Economic Studies* 75(1), 295-316.

Nitsch V., 2000, National Borders and International Trade: Evidence from the European Union, *Cana*dian Journal of Economics 33(4), 1091–1105.

Novy D., 2008, Gravity Redux: Measuring International Trade Costs with Panel Data, University of Warwick, mimeo.

Rauch J.E., 1999, Networks Versus Markets in International Trade, *Journal of International Economics* 48 (1), 7–35.

Rose A.K., 2000, One Money, One Market: Estimating the Effect of Common Currencies on Trade, *Economic Policy* 30, 7-45.

Santos Silva J.M.C. and S. Tenreyro, 2006, The Log of Gravity, *Review of Economics and Statistics* 88(4), 641–658.

Wei S.-J., 1996, Intra-National versus International Trade: How Stubborn Are Nations in Global Integration? *NBER Working Paper* 5531.

Wolf H.C., 2000, Intranational Home Bias in Trade, Review of Economics and Statistics 82(4), 555-563.

3-digit Nace rev.1 Industry	θ^k	$\min\theta^k$	$\max\theta^k$	σ_k	wv^k
Bricks	146.91	5.52	994.94	2.65	4.92
Cement, lime, plaster	109.88	23.21	480.73	2.52	13.36
Stone	54.67	8.29	552.31	2.65	2.14
Fruit, vegetables	41.58	6.11	387.65	2.46	1.03
Articles of concrete, plaster, cement	25.12	2.24	581.78	3.24	3.11
Ceramic tiles, flags	18.15	2.00	266.85	2.65	1.75
Sawmilling, planing of wood	14.75	2.33	106.51	3.36	2.28
Jewellery	14.00	1.95	72.17	3.48	0.01
Other food products	11.31	1.53	$1,\!056.31$	4.81	0.58
Wooden containers	10.27	2.46	87.53	3.99	2.14
Furniture	9.98	1.81	157.52	3.64	0.25
Ceramic goods	9.66	0.86	157.07	3.27	0.71
Prepared animal feeds	9.57	1.04	42.25	3.61	1.75
Glass	8.93	0.80	122.86	3.09	0.84
Builders' carpentry, joinery	8.76	2.74	48.50	3.99	0.51
Processing of iron, steel	6.90	1.71	30.51	3.53	1.10
Publishing	6.66	1.02	78.18	4.88	0.17
Printing	6.39	1.60	21.38	5.58	0.29
Other non-metallic mineral products	5.88	1.23	34.85	2.96	0.57
Structural metal products	5.84	2.27	37.71	4.79	0.36
Other products of wood	5.57	1.79	19.47	4.13	0.38
Tanks, reservoirs	4.48	1.40	22.47	4.76	0.29
Tubes	4.30	1.73	21.12	3.53	0.94
Weapons, ammunition	4.09	2.35	8.80	4.88	0.13
Paper, paperboard	4.00	1.37	12.18	4.64	0.37
Leather clothes	4.00	2.97	5.20	5.66	0.02
Rubber products	3.90	0.80	10.21	3.88	0.26
Fish	3.73	1.35	12.61	4.71	0.33
Pulp, paper, paperboard	3.50	1.36	12.60	4.28	1.71
Other fabricated metal products	3.30	1.24	17.52	4.89	0.39
Wearing apparel	3.02	0.70	15.42	5.66	0.11
Plastic products	2.95	1.29	11.41	5.36	0.28
Miscellaneous manufacturing	2.81	1.41	8.01	4.99	0.13
Grain mill products	2.79	1.65	8.76	5.05	1.69
Musical instruments	2.75	1.47	5.09	4.88	0.05
Cutlery	2.65	1.17	7.08	4.85	0.10
Steam generators	2.59	1.60	6.08	7.87	0.20
Ships, boats	2.57	1.27	4.70	7.40	0.27
Dairy products	2.47	1.57	5.60	6.77	0.52
Luggage, handbags	2.45	1.41	4.20	6.18	0.07

 Table 1: Descriptive Statistics for Individual Industries

(continued on next page)

3-digit Nace rev.1 Industry	$ heta^k$	$\min\theta^k$	$\max\theta^k$	σ_k	wv^k
Soap, detergents	2.30	1.43	10.55	5.74	0.60
Other transport equipment	2.30	1.39	5.05	7.11	0.27
Tobacco	2.29	2.02	2.54	6.62	0.26
Paints, varnishes	2.27	1.41	4.93	6.37	0.44
Basic chemicals	2.22	1.07	6.40	6.09	0.87
Knitted, crocheted articles	2.20	1.12	5.77	6.90	0.04
Electrical equipment	2.14	1.13	4.70	5.94	0.06
Lighting equipment	2.11	1.19	4.26	5.17	0.09
Railway locomotives	2.10	1.31	4.74	7.40	0.13
Insulated wire, cable	2.10	1.26	3.27	5.88	0.19
Knitted, crocheted fabrics	2.07	1.51	3.35	6.90	0.10
Other chemical products	2.04	1.00	6.01	6.46	0.30
Electricity distribution apparatus	2.01	1.33	2.80	5.88	0.04
Sports goods	2.00	0.68	3.79	4.88	0.12
Domestic appliances	1.94	1.04	4.05	5.75	0.18
Made-up textile articles	1.89	1.12	2.93	7.46	0.11
Man-made fibres	1.87	1.41	2.51	6.59	0.42
Footwear	1.86	1.12	3.25	7.22	0.04
Medical equipment	1.84	1.43	2.63	6.00	0.03
Other general purpose machinery	1.84	1.25	4.28	7.00	0.13
Agricultural, forestry machinery	1.83	1.19	3.03	8.36	0.17
Parts for motor vehicles	1.75	1.06	4.50	7.28	0.17
Electronic valves, tubes	1.74	1.04	2.84	5.88	0.02
Other textiles	1.68	0.96	3.13	7.82	0.17
Other special purpose machinery	1.66	1.09	3.25	8.17	0.08
Pesticides	1.66	1.22	2.40	6.75	0.28
Tanning, dressing of leather	1.62	1.14	3.35	8.92	0.08
Pharmaceuticals	1.62	1.11	3.39	9.05	0.05
Machinery	1.57	0.88	2.86	7.21	0.09
Electronic motors	1.56	1.24	2.06	7.02	0.09
Dressing, dyeing of fur	1.53	1.19	1.97	8.09	0.02
Motorcycles, bicycles	1.52	1.08	1.93	7.11	0.10
Optical instruments	1.52	1.21	2.10	7.70	0.01
Accumulators	1.48	1.12	1.87	5.88	0.32
Games, toys	1.41	1.11	1.72	4.88	0.10
Motor vehicles	1.40	1.21	1.61	7.25	0.12
Aircraft, spacecraft	1.37	1.21	1.60	7.55	0.00
Office machinery, computers	1.15	1.04	1.24	10.94	0.02

Table 1 (continued)

Office machinery, computers1.151.041.2410.940.02Notes: Authors' calculations. For θ^k , σ_k and wv^k , the numbers reported are averages for each3-digit industry. wv^k is in units of kg/Euro.

	θ_i	$\min\theta_i$	$\max\theta_i$
Ireland	9.47	1.208	261.49
Spain	9.02	0.677	1056.30
Denmark	8.80	1.138	782.81
Finland	8.61	1.195	552.31
Italy	7.75	0.701	1,056.30
United Kingdom	7.73	0.999	994.94
Austria	6.69	0.927	154.63
Portugal	6.50	0.677	414.15
Germany	5.67	0.879	480.73
Netherlands	5.48	0.970	123.36
France	5.28	0.701	414.15

 Table 2: Descriptive Statistics for Individual Countries

Notes: Authors' calculations. The numbers reported for θ_i are averages for each country across industries.

	(1)	(2)	(3)	(4)	(5)	(6)
Geography/Transport Costs						
$\ln D_{ij}$	$0.583 \atop (28.378) a$	$0.515 \atop {}^a_{(23.296)}$	$0.530 \\ {}^{a}_{(24.799)}$	$0.483 \\ {}^{a}_{(21.420)}$	$0.495 \atop (22.074)^a$	${0.506\atop_{(22.806)}}^a$
$\ln\left(D_{ii}\times D_{jj}\right)$	${-0.930}^{a}_{(-11.973)}$	$\substack{-0.945\ ^{a}}_{(-11.381)}$	$-0.990^{\ a}_{\ (-12.247)}$	${-0.923}^{a}_{(-11.323)}$	${}^{-0.928}_{\scriptscriptstyle (-11.672)}{}^{a}$	$-0.969^{\ a}_{\ (-12.354)}$
Adj_{ij}	${-0.063^a}_{(-2.810)}$	${-0.051}^{b}_{(-2.218)}$	$-0.044^{c}_{(-1.939)}$	-0.068^{a}	${-0.061}^{a}_{(-2.625)}$	${-0.055}^b_{(-2.428)}$
$Lang_{ij}$	${-0.201^a}_{(-4.854)}$	${-0.355^a\atop_{(-8.109)}}$	${-0.353^a}_{(-7.995)}$	$-0.363^{a}_{(-8.186)}$	${-0.369^a}_{(-8.432)}$	${-0.368^a}_{(-8.331)}$
$\ln c f o b_t^k$	$0.030^{a}_{(5.464)}$	$0.030^{a}_{(6.152)}$	$0.031^{a}_{(6.294)}$	$0.030^{a}_{(6.146)}$	$0.015^{a}_{(3.291)}$	$0.016^{a}_{(3.400)}$
$\ln w v_t^k$	$0.719^a_{(5.973)}$	${0.631 \atop (6.414)}^a$	${0.642}^{a}_{(6.459)}$	$0.648^{a}_{(6.698)}$	$0.664^{a}_{(6.833)}$	${0.673 \atop (6.854)}^{a}$
Policy Variables						
FI, AT_{ij}	_	$0.199 \\ {}^{a}_{(11.215)}$	$0.187 \\ {}^{a}_{(10.666)}$	$0.233 \\ {}^{a}_{(12.376)}$	$0.228\ ^a_{(12.331)}$	$0.220\ ^{a}_{(11.851)}$
$noEURO_{ij,t}$	—	${0.042}^{a}_{(3.108)}$	${0.032}^{b}_{(2.388)}$	$0.040^{a}_{(2.906)}$	$0.036^{a}_{(2.704)}$	$0.027^{b}_{(2.072)}$
$Schengen_{ij,t}$	_	${-0.137^a}_{(-5.443)}$	${-0.118^a}_{(-4.774)}$	${-0.135^a}_{(-4.914)}$	${-0.132^a}_{(-4.954)}$	${-0.113^a}_{(-4.302)}$
$\ln TBT_{ij}^k$	—	$0.142^{a}_{(2.822)}$	_	$0.146^{a}_{(2.721)}$	$0.131^{b}_{(2.434)}$	_
TBT^k	—	_	$0.057^{b}_{(2.156)}$	_	_	$0.056^{b}_{(1.967)}$
$\ln Proc_{ij,t}^k$	_	$-0.573^{c}_{(-1.886)}$	${-0.528}^{c}_{(-1.738)}$	${-0.963}^a_{(-2.729)}$	${\displaystyle -0.917^{b}} \atop {\scriptstyle (-2.564)}$	${-0.892^b}_{(-2.495)}$
$\ln VAT_{ij}^k$	—	$-2.282^{a}_{(-5.741)}$	$-2.290^{a}_{(-5.744)}$	$-2.259^{a}_{(-5.693)}$	-1.910^a $_{(-5.122)}$	$-1.931^{a}_{(-5.138)}$
Other Costs						
$\ln Prod_{ij,t-1}^k$	—	_	_	$-0.143^{a}_{(-4.453)}$	$-0.144^{a}_{(-4.581)}$	$-0.156^{a}_{(-5.120)}$
$Zeros^k_{ij,t}$	_	_	_	$2.277^{a}_{(9.913)}$	$2.210^{a}_{(9.849)}$	$2.193^{a}_{(9.511)}$
Controls						
$KS^k_{ij,t}$	—	_	_	_	0.054 (1.465)	$\underset{(1.637)}{0.063}$
$\ln Goods^k$	—	_	_	_	${-0.112^a}_{(-9.653)}$	${-0.112^a}_{(-9.633)}$
N	15,040	15,040	15,040	12,032	12,032	12,032
$\mathrm{Adj}\text{-}R^2$	0.791	0.802	0.801	0.804	0.810	0.810

 Table 3: The Determinants of EU Trade Integration

Notes: The dependent variable is $\ln \theta_{ij,t}^k$. Robust standard errors are adjusted for clustering at the 4-digit Nace rev.1 level in each country pair (3,008 clusters). Year and 3-digit industry fixed effects are included in all regressions. The sample period is 1999-2003. *t*-statistics in parentheses. Constant terms are included but not reported. ^{*a*}, ^{*b*} and ^{*c*} indicate significance at 1, 5 and 10 percent levels, respectively.

	(1)	(2)
Geography/Transport Costs	24.6%	25.0%
$\ln D_{ij}$	5.4%	5.6%
$\ln\left(D_{ii}\times D_{jj}\right)$	0.7%	0.8%
Adj_{ij}	0.6%	0.5%
$Lang_{ij}$	0.5%	0.5%
$\ln c f o b_t^k$	0.7%	0.7%
$\ln w v_t^k$	16.7%	16.9%
Policy Variables	7.6%	6.8%
FI, AT_{ij}	1.4%	1.4%
$noEURO_{ij,t}$	0.01%	0.01%
$Schengen_{ij,t}$	0.2%	0.2%
$\ln TBT_{ij}^k$	4.5%	_
TBT^k	_	3.7%
$\ln Proc_{ij,t}^k$	0.6%	0.6%
$\ln VAT_{ij}^k$	0.8%	0.8%
Other Costs	0.3%	0.4%
$\ln Prod_{ij,t-1}^k$	0.1%	0.1%
$Zeros_{ij,t}^k$	0.3%	0.3%
Controls	5.6%	5.5%
$KS^k_{ij,t}$	0.1%	0.1%
$\ln Goods^k$	5.4%	5.4%
3-digit industry fixed effects	42.8%	43.2%
Year fixed effects	0.01%	0.01%
Variation explained	80.8%	80.8%
Residual	19.2%	19.2%
Sum	100%	100%
N	12,032	12,032

Table 4: Variance Decomposition

Notes: The variance decompositions are calculated according to Fields (2003). The contribution of each explanatory variable x_m to the total variance of $\theta_{ij,t}^k$ is given by $c_m = \beta_m cov \left(x_m, \theta_{ij,t}^k\right) / var \left(\theta_{ij,t}^k\right)$ where β_m is the partial regression coefficient of $\theta_{ij,t}^k$ on the explanatory variable x_m (holding all other explanatory variables constant). The decompositions in Columns (1) and (2) correspond to regressions (5) and (6) in Table 3. The contributions sum to 100 percent.

"Naive" Gravity	(1)	(2)	(3)	(4)
$\ln D_{ij}$	$-3.589^{\ a}$	$-3.741^{\ a}$	$-3.629^{\ a}$	-3.757 ^a
$\ln\left(y_i^ky_j^k\right)$	(-31.563) 1.226^{a} (57.064)	(-46.554) 1.525^{a} (68.313)	$^{(-30.342)}$	(-45.862)
$\ln y_i^k$	-	_	1.271^{a}	1.546^{a}
$\ln y_j^k$	_	_	(30.217) 1.184 ^{<i>a</i>} (30.617)	(43.708) 1.507^{a} (53.293)
Distance coefficients	Same $\forall k$	k-specific	Same $\forall k$	k-specific
$\operatorname{Adj-}R^2$	0.602	0.849	0.602	0.849
Micro-Founded Gravity	(1)	(2)	(3)	(4)
$\ln D_{ij}$	$-4.514^{\ a}_{\ (-34.104)}$	$-5.131^{\ a}_{\ (-52.663)}$	$-4.541^{\ a}_{\ (-32.669)}$	$-5.045^{\ a}_{\ (-53.067)}$
$\ln\left(D_{ii}\times D_{jj}\right)$	$11.551^{a}_{(19.551)}$	$7.584 \stackrel{a}{_{(17.071)}}$	_	_
$\ln D_{ii}$	_	_	$2.022\ ^{a}_{(13.343)}$	$1.113^{a}_{(6.612)}$
$\ln D_{jj}$	_	_	$2.223 \atop (12.676) ^a$	${1.659\atop_{(12.387)}}^a$
$\ln\left(x_{ii}^kx_{jj}^k ight)$	${0.750\atop_{(31.486)}}^{a}$	${1.010\atop (41.879)}^a$	_	_
$\ln x_{ii}^k$	_	_	$0.789^{\ a}_{(16.522)}$	${1.048\atop_{(27.501)}}^{a}$
$\ln x_{jj}^k$	_	_	$0.712^{\ a}_{(15.942)}$	$0.967 \\ {}^{a}_{(28.263)}$
Distance coefficients	Same $\forall k$	k-specific	Same $\forall k$	k-specific
$\operatorname{Adj-}R^2$	0.510	0.842	0.510	0.852
Micro-Founded Gravity with FE	(1)	(2)	(3)	(4)
$\ln D_{ij}$	$-4.204^{\ a}_{\ (-66.739)}$	$-4.653^{\ a}_{\ (-15.537)}$	_	_
$\ln\left(y_i^k y_j^k\right)$	$1.388 \atop (39.887) ^{a}$	$rac{1.462}{\scriptscriptstyle (40.914)}^{a}$	_	_
Distance coefficients	Same $\forall k$	k-specific	_	_
$\operatorname{Adj-}R^2$	0.581	0.639	_	_

Table 5: The Gravity Equation

Notes: The upper, middle and lower panels refer to the naive and the two micro-founded gravity equations, estimated according to (17), (18) and (19) in the text. In Columns (1) and (3), the coefficients on bilateral and domestic distances are constrained to be the same across industries. In Columns (2) and (4), they are allowed to vary across industries, in which case we report the average of the elasticities estimated across industries. In all cases the number of observations is equal to 15,040. The sample period is 1999-2003. Constant terms are included but not reported. *t*-statistics in parentheses. ^{*a*}, ^{*b*} and ^{*c*} indicate significance at 1, 5 and 10 percent levels, respectively.

	Trade growth	Factor (1)	Factor (2)	N
	1999-2003,	Contribution of	Contribution of	
	average, in $\%$	output growth	trade integration	
Full sample	7.96	42%	58%	12,032
Geography/Transport Costs				
[†] Long D_{ij}	9.13	63%	$\mathbf{37\%}$	6,452
[†] Short D_{ij}	6.78	21%	$\mathbf{79\%}$	5,580
$Adj_{ij} = 0$	7.65	47%	$\mathbf{53\%}$	9,300
$Adj_{ij} = 1$	8.91	$\mathbf{28\%}$	$\mathbf{72\%}$	2,732
[†] High $cfob_{1999}^k$	7.50	70%	30%	4,704
[†] Low $cfob_{1999}^k$	8.24	$\mathbf{26\%}$	$\mathbf{74\%}$	7,328
[†] High wv_{1999}^k	7.07	54%	46%	3,744
[†] Low wv_{1999}^k	8.36	$\mathbf{37\%}$	$\mathbf{63\%}$	8,288
$Lang_{ij} = 0$	8.06	43%	57%	11,704
$Lang_{ij} = 1$	4.83	$\mathbf{24\%}$	$\mathbf{76\%}$	328
Policy Variables				
FI, AT_{ij}	9.06	36%	$\mathbf{64\%}$	3,432
Rest of sample	7.58	44%	$\mathbf{56\%}$	8,600
$noEURO_{ij}$ (from 2002)	0.32	85%	15%	1,796
$EURO_{ij}$ (from 2002)	9.98	21%	$\mathbf{79\%}$	4,132
$Schengen_{ij,t} = 0$	6.56	144%	-44%	270
$Schengen_{ij,t} = 0.5$	6.89	79%	$\mathbf{21\%}$	$3,\!455$
$Schengen_{ij,t} = 1$	8.45	$\mathbf{24\%}$	$\mathbf{76\%}$	8,307
[†] High TBT_{ij}^k	7.41	17%	83%	4,232
[†] Low TBT_{ij}^{k}	8.25	55%	45%	7,800
† Low $Proc_{ij,1999}$	7.10	24%	76%	5,812
[†] High $Proc_{ij,1999}$	8.79	60%	40%	6,220
Productivity				
[†] Low $Prod_{ij,1999}^k$	6.88	76%	$\mathbf{24\%}$	5,848
[†] High $Prod_{ij,1999}^{k}$	8.93	12%	88%	6,184

Table 6: Decomposition of the Growth in Trade

Notes: Decompositions are obtained according to equation (23) in the text. The numbers reported for trade growth and for factors (1) and (2) are weighted averages of the individual observations for each country pair and industry where the weights are the products of sectoral bilateral exports in the initial year 1999. [†] denotes that the sample is split at the mean value of the variable in question in the initial year 1999. Mean values are equal to $D_{ij} = 1, 161$ kilometers, $cfob_{1999}^k = 4.43$, $wv_{1999}^k = 1.41$ kg/Euro, $TBT_{ij}^k = 3.65$, $Proc_{ij,1999} = 1.12$ and $Prod_{ij,1999}^k = 41,060$ Euros/employee.



Figure 1: Trade frictions $\ln \theta_t,$ 1999-2003, average across countries and industries.



Figure 2: Annualized growth rates per industry, 1999-2003, see regression (20).

 \bullet indicates significance at the 10 percent level. \times indicates insignificantly different from zero.

Appendix A

	(1)	(2)	(3)	(4)	(5)	(6)
Geography/Transport Costs						
$\ln D_{ij}$	$0.786^{a}_{(7.221)}$	—	$0.492 \\ {}^{a}_{(21.932)}$	${0.465\atop_{(22.586)}}^a$	$0.495 \atop (21.828) ^a$	${3.785\atop_{(28.313)}}^a$
$\ln\left(D_{ii}\times D_{jj}\right)$	$-1.112^{a}_{(-3.151)}$	_	${-0.925}^{a}_{(-11.642)}$	$\substack{-0.873\ ^{a}}_{(-11.970)}$	${-0.927}^{a}_{(-11.556)}$	$-7.779^{\ a}_{\ (-16.284)}$
Adj_{ij}	${-0.449^a}_{(-3.629)}$	_	${-0.063^a}_{(-2.704)}$	${-0.060^a} \atop {\scriptstyle (-2.852)}$	${-0.061}^a_{(-2.603)}$	-0.097 (-0.811)
$Lang_{ij}$	-0.159 $_{(-1.145)}$	_	${-0.368^a} \atop {\scriptstyle (-8.423)}$	${-0.357^a}_{(-8.612)}$	${-0.369^a} \atop {\scriptstyle (-8.363)}$	$-2.590^{\ a}_{\ (-11.979)}$
$\ln c f o b_t^k$	$0.067^{a}_{(3.223)}$	${0.017}^{a}_{(3.596)}$	${0.015}^{a}_{(3.174)}$	$0.013^{a}_{(3.106)}$	$0.018^{a}_{(2.615)}$	$0.150^{a}_{(5.350)}$
$\ln w v_t^k$	$0.606 \atop (3.814)^a$	$0.670^{a}_{(7.007)}$	$0.665 \atop_{(6.854)}^a$	$0.648^{a}_{(7.255)}$	$0.724^{a}_{(6.826)}$	${1.961 \atop (6.120)}^{a}$
Policy Variables						
FI, AT_{ij}	$0.406^{a}_{(5.146)}$	_	$0.227 \\ {}^{a}_{(12.253)}$	$0.215 \\ {}^{a}_{(12.390)}$	$0.229^{\ a}_{(12.212)}$	${1.827\atop_{(16.812)}}^a$
$noEURO_{ij,t}$	$0.248^{a}_{(3.212)}$	_	$0.034^{b}_{(2.549)}$	$0.032^{a}_{(2.675)}$	$0.037^{a}_{(2.628)}$	$\underset{(0.950)}{0.077}$
$Schengen_{ij,t}$	-0.449^{a} $_{(-3.265)}$	—	$-0.130^{a}_{(-4.928)}$	${-0.119^a}_{(-4.930)}$	$-0.134^{a}_{(-4.937)}$	${-0.577^a}_{(-3.696)}$
$\ln TBT^k_{ij}$	$1.110^{a}_{(4.430)}$	$0.105^{c}_{(1.876)}$	$0.133^{b}_{(2.473)}$	$0.127^{a}_{(2.664)}$	${0.125 \atop (2.346)}^{b}$	$\underset{(-0.430)}{-0.057}$
$\ln Proc_{ij,t}^k$	${-3.320^a}_{(-3.245)}$	${-1.377^a}_{(-3.789)}$	${-0.899^b}_{(-2.521)}$	${-0.779^b}_{(-2.409)}$	${-1.279^a}_{(-2.752)}$	$-5.717^{b}_{(-1.967)}$
$\ln VAT^k_{ij}$	${-4.103^b}_{(-2.468)}$	$-2.520^{a}_{(-5.573)}$	-1.848^{a} $^{(-4.979)}$	${-1.502^a}_{(-4.279)}$	$-1.917^{a}_{(-5.063)}$	$-13.274^{a}_{(-5.579)}$
Other Costs						
$\ln Prod_{ij,t-1}^k$	${-0.297^a} \atop {\scriptstyle (-2.784)}$	-0.109^a $_{(-2.615)}$	$-0.144^{a}_{(-4.574)}$	${-0.134^a}_{(-4.501)}$	${-0.152^a}_{(-4.720)}$	-1.638^{a} (-9.872)
$Zeros_{ij,t}^k$	$2.145_{(13.827)}^{a}$	$2.249^{\ a}_{(12.775)}$	_	$2.287^{a}_{(9.872)}$	$2.214^{a}_{(9.853)}$	${13.245^a}_{(11.914)}$
Controls						
$KS_{ij,t}^k$	$0.318^{c}_{(1.740)}$	$\underset{(1.027)}{0.042}$	$\underset{(1.459)}{0.054}$	$\underset{(1.610)}{0.055}$	$\underset{(1.467)}{0.055}$	$\underset{(1.599)}{0.280}$
$\ln Goods^k$	${-0.181}^a_{(-4.381)}$	$\begin{array}{c} -0.112^{a} \\ {}_{(-9.888)} \end{array}$	$\begin{array}{c} -0.110^{a} \\ {}_{(-9.487)} \end{array}$	${-0.102^a}_{(-9.680)}$	$\begin{array}{c} -0.111^a \\ {}_{(-9.465)} \end{array}$	$-1.205^{\ a}_{\ (-15.924)}$
Estimator	PPML	OLS	OLS	OLS	OLS	OLS
Fixed Effects	t, K	$t\times ij,K$	t, K	t, K	$t \times K$	t, K
Weighted	No	No	No	Yes	No	No
Sample	Full	Full	Excl. zeros	Full	Full	Full
N	12,032	12,032	12,016	12,032	12,032	12,032
Adj - R^2	_	0.821	0.810	0.812	0.812	0.693

 Table A1: The Determinants of EU Trade Integration – Robustness

Notes: In (1) the dependent variable is $\theta_{ij,t}^k$ and in (2) to (5) the dependent variable is $\ln \theta_{ij,t}^k$. In (6), the dependent variable is the (log) inverse *phi*-ness of trade. Poisson Pseudo-Maximum Likelihood (PPML) estimation in (1); fixed effects OLS regressions in (2) to (6). Robust standard errors are adjusted for clustering at the 4-digit Nace rev.1 level in each country pair (3,008 clusters). The sample period is 1999-2003. Weighted regression in (4) where the weights are the product of sectoral bilateral exports in the initial year 1999. *t*-statistics in parentheses. Constant terms are included but not reported. ^a, ^b and ^c indicate significance at 1, 5 and 10 percent levels, respectively.

	1999	2000	2001	2002	2003
Geography/Transport Costs					
$\ln D_{ij}$	$0.515^{\ a}_{\ (22.108)}$	$0.475 \\ {}^{a}_{(19.175)}$	$0.515 \\ {}^{a}_{(20.834)}$	$0.493 \\ {}^{a}_{(20.453)}$	$0.498 \\ {}^{a}_{(20.306)}$
$\ln\left(D_{ii}\times D_{jj}\right)$	$\left. {\begin{array}{*{20}c} -0.857 \ ^{a} \\ {}_{(-10.064)} \end{array} } \right.$	$\substack{-0.955 \\ \scriptstyle (-10.932)}^{a}$	${-0.988 \over (-10.942)}^a$	${-0.906}^{a}_{(-11.068)}$	${-0.836}^a_{(-9.928)}$
Adj_{ij}	$\left \begin{array}{c} -0.050^b \\ _{(-2.146)} \end{array} \right $	-0.068^{a} (-2.860)	-0.057^b $_{(-2.225)}$	-0.054^{b}	-0.056^{b} (-2.180)
$Lang_{ij}$	$\left \begin{array}{c} -0.351^{a} \\ _{(-7.436)} \end{array} \right $	$-0.357^{a}_{(-7.970)}$	-0.376^{a}	-0.383^{a}	${-0.377^a}_{(-7.701)}$
$\ln c f o b_t^k$	0.031^{b} (2.109)	0.015 (1.347)	0.004 (0.301)	0.002 (0.204)	0.040^{a} (3.292)
$\ln w v_t^k$	$0.793^{a}_{(7.132)}$	$0.928^{a}_{(7.775)}$	$0.743^{a}_{(6.202)}$	$0.733^{a}_{(6.011)}$	$0.541^{a}_{(4.804)}$
Policy Variables					
FI, AT_{ij}	0.179^{a} (9.504)	$0.219\ ^{a}_{(11.122)}$	${0.224\atop_{(10.800)}}^{a}$	${0.237\atop_{(11.875)}}^{a}$	${0.246\atop_{(12.257)}}^a$
$noEURO_{ij,t}$	_	_	_	0.029 (1.353)	$0.061^{a}_{(2.690)}$
$Schengen_{ij,t}$	$\left \begin{array}{c} -0.113^{a} \\ (-3.751) \end{array}\right $	$-0.123^{a}_{(-4.136)}$	-0.161^{a}	-0.150^{a}	-0.134^{a}
$\ln TBT_{ij}^k$	0.098^{b} (2.393)	0.059 (1.572)	$0.125^{b}_{(2.162)}$	$0.142^{a}_{(2.595)}$	$0.170^{a}_{(2.589)}$
$\ln Proc_{ij,t}^k$	-0.113 (-0.128)	-0.728 (-1.342)	$-1.243^{c}_{(-1.925)}$	-0.752 (-1.034)	-1.866^{a} (-4.520)
$\ln VAT_{ij}^k$	-1.817^{a}	$-2.269^{a}_{(-5.226)}$	-1.847^{a}	-1.407^{a}	-2.154^{a}
Other Costs					
$\ln Prod_{ij,t-1}^k$	_	-0.198^{a}	-0.144^{a}	-0.108^{a}	$-0.163^{a}_{(-4.249)}$
$Zeros^k_{ij,t}$	_	_	_	$2.239^{a}_{(7.733)}$	$2.205^{\ a}_{(11.095)}$
Controls					
$KS^k_{ij,t}$	-0.060 (-1.269)	-0.044 (-1.005)	$\underset{(1.744)}{0.076}^{c}$	$0.083^{c}_{(1.827)}$	${0.137}^{b}_{(2.412)}$
$\ln Goods^k$	$\left \begin{array}{c} -0.096^{a} \\ {}_{(-7.040)} \end{array} \right $	$-0.114^{a}_{(-8.656)}$	${-0.121}^{a}_{(-9.038)}$	${-0.111}^{a}_{(-9.161)}$	-0.098^a $_{(-7.577)}$
N	3,008	3,008	3,008	3,008	3,008
Adj - R^2	0.818	0.814	0.811	0.817	0.813

Table A2: The Determinants of EU Trade Integration: Cross-sectional samples

Notes: The dependent variable is $\ln \theta_{ij,t}^k$. Robust standard errors are adjusted for clustering at the 4-digit Nace rev.1 level in each country pair (3,008 clusters). 3-digit industry fixed effects are included in all regressions. The sample period is 1999-2003. *t*-statistics in parentheses. Constant terms are included but not reported. ^{*a*}, ^{*b*} and ^{*c*} indicate significance at 1, 5 and 10 percent levels, respectively.

Appendix B: Technical Appendix

As discussed in Section 2.4, our approach is consistent with a number of leading trade theories. This appendix provides the details. For simplicity we drop the industry index k so that we can write the trade integration measure in equation (14) as

$$\theta_{ij} = \left(\frac{x_{ii}x_{jj}}{x_{ij}x_{ji}}\right)^{\frac{1}{2(\sigma-1)}} = \left(\frac{t_{ij}t_{ji}}{t_{ii}t_{jj}}\right)^{\frac{1}{2}}$$
(24)

We use the notation from our model when referring to expressions from the papers below.

Eaton and Kortum (2002) In the Ricardian model by Eaton and Kortum (2002), productivity in each country is drawn from a Fréchet distribution that has two parameters, T_i and ζ . T_i determines the location of the productivity distribution for country *i*, with a high T_i denoting high overall productivity. $\zeta > 1$ denotes the variation of productivity across goods and is treated as common across countries, with a high ζ denoting little variation. Costinot and Komunjer (2007) generalize the model to multiple industries. The model yields a gravity equation for an aggregate of homogeneous goods whose structure is similar to equation (7). It is given by

$$\frac{x_{ij}}{x_j} = \frac{T_i (c_i t_{ij})^{-\zeta}}{\sum_{i=1}^J T_i (c_i t_{ij})^{-\zeta}}$$

where c_i denotes the input cost in country *i*. We are interested in the trade cost parameters. T_i and c_i are unobservable but cancel out once the ratio of domestic over bilateral trade flows is formed as in equation (24). This yields

$$\theta_{ij}^{EK} = \left(\frac{x_{ii}x_{jj}}{x_{ij}x_{ji}}\right)^{\frac{1}{2\zeta}} = \left(\frac{t_{ij}t_{ji}}{t_{ii}t_{jj}}\right)^{\frac{1}{2}}$$
(25)

Comparing equations (24) and (25), it is obvious that $\theta_{ij}^{EK} = \theta_{ij}$ if $\zeta = \sigma - 1$. For more details on the comparison of Armington-type and Ricardian models, see Eaton and Kortum (2002, footnote 20) and Anderson and van Wincoop (2004, pp. 709-710).

Chaney (2008) Chaney (2008) builds on the seminal paper by Melitz (2003) and derives a gravity equation based on a model with heterogeneous productivity across firms and fixed costs of exporting. In contrast to standard trade models, the two assumptions of heterogeneous firms and fixed costs of exporting introduce an extensive margin of trade. Exporters do not only vary the size of shipments in response to trade cost changes (the intensive margin), but the set of exporters also changes (the extensive margin). Chaney derives the following industry-level gravity equation

$$x_{ij} = \mu \frac{y_i y_j}{y} \left(\frac{w_i t_{ij}}{\lambda_j}\right)^{-\gamma} (f_{ij})^{-\left(\frac{\gamma}{\sigma-1}-1\right)}$$

where μ is the weight of the industry in the consumers' utility function, y_i is total income of country i, y is world income, w_i is workers' productivity in country i, λ_j is a remoteness variable akin to multilateral resistance and f_{ij} are the fixed costs of exporting from country i to j. γ is the shape parameter of the Pareto distribution, with a high γ denoting a low degree of heterogeneity and $\gamma > \sigma - 1$. Forming the ratio of domestic over bilateral trade flows yields

$$\theta_{ij}^{Ch} = \left(\frac{x_{ii}x_{jj}}{x_{ij}x_{ji}}\right)^{\frac{1}{2\gamma}} = \left(\frac{t_{ij}t_{ji}}{t_{ii}t_{jj}}\right)^{\frac{1}{2}} \left(\frac{f_{ij}f_{ji}}{f_{ii}f_{jj}}\right)^{\frac{1}{2}\left(\frac{1}{\sigma-1}-\frac{1}{\gamma}\right)}$$
(26)

The trade integration measure θ_{ij}^{Ch} is a function of both variable and fixed trade costs. For details on welfare implications of the above models, see Arkolakis, Demidova, Klenow and Rodríguez-Clare (2008).

Melitz and Ottaviano (2008) Melitz and Ottaviano (2008) also model heterogeneous firms. But in contrast to Melitz (2003) and Chaney (2008), firms face fixed costs of market entry f_E that can be interpreted as product development and production start-up costs. When exporting, the firms only face variable trade costs and no fixed costs of exporting. The model is based on non-CES preferences that give rise to endogenous markups. More specifically, markups tend to be low in large markets with many competitors. Their multiple country model leads to the following gravity equation

$$x_{ij} = \frac{1}{2\delta(\gamma+2)} N_E^i \psi^i L^j \left(c_d^j\right)^{\gamma+2} (t_{ij})^{-\gamma}$$

where δ is a parameter from the utility function that indicates the degree of product differentiation (a higher δ means a higher degree of differentiation). N_E^i is the number of entrants in country *i*. ψ^i is an index of comparative advantage in technology. A high ψ^i means that entrants in country *i* have a high chance of obtaining good productivity draws. L^j denotes the number of consumers in country *j*. c_d^j is the marginal cost cut-off above which domestic firms in country *j* do not produce. The intuition is that tougher competition in country *j*, reflected by a lower c_d^j , makes it harder for exporters from *i* to break into that market. Forming the ratio of domestic over bilateral trade flows yields

$$\theta_{ij}^{MO} = \left(\frac{x_{ii}x_{jj}}{x_{ij}x_{ji}}\right)^{\frac{1}{2\gamma}} = \left(\frac{t_{ij}t_{ji}}{t_{ii}t_{jj}}\right)^{\frac{1}{2}}$$
(27)

Fixed costs do not enter the trade integration measure θ_{ij}^{MO} because all firms face identical entry costs f_E and no fixed costs of exporting. Variable trade costs are sufficient to induce selection into export markets because of bounded non-CES marginal utility.

Summary The four measures θ_{ij} , θ_{ij}^{EK} , θ_{ij}^{Ch} and θ_{ij}^{MO} have in common that they scale the ratio of domestic over bilateral trade flows by parameters that indicate a particular form of heterogeneity. A low σ in equation (24) indicates a high degree of differentiation across products; a low ζ in equation (25) indicates a high variation of productivity across countries; and a low γ in equations (26) and (27) indicates a high degree of firm heterogeneity. For given trade flows, higher heterogeneity in equations (24)-(27) implies higher trade frictions. Intuitively, suppose we observe two industries with the same ratio of domestic over bilateral trade but one industry is characterized by a higher degree of heterogeneity. With higher heterogeneity countries produce goods that are more different from each other or reflect larger differences in comparative advantage. Thus, consumers have a larger incentive to trade. The fact that consumers do not trade more implies that trade frictions must be higher in that industry.

Appendix C: Data Appendix

Trade data, weight-to-value and c.i.f./f.o.b. Bilateral and total export and import trade flows (thousand Euros), and their corresponding weight (tons) are used for 11 EU countries between 1999 and 2003 at the 4-digit Nace rev.1 level of manufacturing industries. *Source: Eurostat.*

In Section 4.1, the bilateral weight-to-value ratio of exports (kilograms per Euro exported) is calculated separately for the two partners in each industry and in each year, and is then averaged across all partners. In Section 4.3, the bilateral weight-to-value ratio of exports is calculated for each country, industry and year. In both cases, we calculate the log of one plus weight-to-value as weight-to-value often takes on values much smaller than one.

The ratio between bilateral import ("Costs, Insurance and Freight," *c.i.f.*) and export ("Free On Board," *f.o.b.*) flows is calculated separately for the two partners in each industry and in each year, and is then averaged across all partners, dropping the few cases where the ratio is smaller than one. We then use the log of the ratio.

Output Gross value of output (million Euros) at the 4-digit Nace rev.1 level. Source: Eurostat's New Cronos database.

Elasticities of Substitution We draw on estimates by Hummels (2001a) who uses data on bilateral trade flows, import tariffs and transport costs to estimate elasticities of substitution at the 2-digit SITC rev.3 level of industries. We use his OLS estimates (he also reports IV estimates which yield higher values for the elasticities). The elasticities are converted using tables of correspondence from the SITC rev.3 to the ISIC rev.3, and then from the ISIC rev.3 to the Nace rev.1. In the few cases where the Nace rev.1 level industries have to be matched with several SITC industries, we calculate the average across SITC industries. Ideally, one would compute a weighted average where the weights are given by the share of each SITC rev.3 industry into each Nace rev.1 industry grouping, but this information is not available. Among the 62 elasticities, 4 are not significantly different from zero so we set them to missing. *Source: Hummels (2001a)*.

Gravity Variables Dummies for sharing a common land border and for sharing a common (official) language. Source: Centre d'Etudes Prospectives et d'Informations Internationales (CEPII). International and domestic distances, which are weighted averages of the distances between regions using GDP shares as weights. Source: Chen (2004).

Schengen Agreement For each country, a dummy is equal to one in the years in which the provisions of the Schengen Agreement are fully implemented. We then take the average between the two partners. The resulting variable takes on values of zero, one half and one. The years of first implementation are chosen depending on the month in which the Agreement went into force: Austria (1998-2004), Denmark (2001-2004), Finland (2001-2004), France (1995-2004), Germany (1995-2004), Italy (1998-2004), Netherlands (1995-2004), Portugal (1995-2004), Spain (1995-2004) while Ireland and the United Kingdom have not yet fully implemented the Agreement.

Technical Barriers to Trade We use two sources. The European Commission's Eurobarometer reports opinions and experiences of European managers about the Single Market. A total of 4,900 managers at companies were interviewed by telephone in early 2006, the sample of companies being selected according to the size of countries and of companies, and the industry of activity. We use the answer to the question: "Could you tell me whether you consider that for your company it is very important, rather important, rather unimportant or not important at all that future Single Market Policy tackles the question of *removing remaining technical barriers to trade in goods?*" For each country, we group the answers from all managers who replied that TBTs are indeed an important issue, and use the percentage so obtained as a <u>country-specific</u> indication on the relevance of TBTs. Source: European Commission (2006).

To capture the <u>sectoral</u> relevance of TBTs, industries are classified on a five-point scale according to the effectiveness of different measures undertaken by the Single Market Programme to eliminate TBTs: measures are successful and all significant barriers are removed (value of 1), measures are implemented and function well but some barriers remain (value of 2), measures are adopted but with implementation or transitional problems still to be overcome (value of 3), measures are proposed or implemented but not effective or with operating problems (value of 4), and no solution has been adopted (value of 5). Some industries are also identified as being not affected by TBTs prior to European integration (and are given a value of 0). Our industry-specific qualitative variable takes on values between 0 and 5, with larger values indicating a lack of market integration due to persisting TBTs. *Source: European Commission (1998)*.

We then interact the log of one plus the average across partners of the share of managers replying that TBTs are important with one plus the industry-specific variable on TBTs.

Public Procurement We use two sources. For each <u>country</u>, we use the proportion of public procurement contracts advertised each year in the *Official Journal* of the EU. *Source: Eurostat*. The <u>industries</u> strongly affected by public procurement in Europe are identified at the Nace 70 level, are then converted at the Nace rev.1 level and identified by a dummy variable. *Source: Davies and Lyons (1996)*. We calculate the log of one plus the average across partners of the proportion of public procurement contracts advertised in the *Official Journal* of the EU, which we then interact with the sectoral dummy.

Value-Added Taxes For each country, we use the standard VAT rate and replace it by the reduced VAT rates that apply to some categories of goods, as of January 1^{st} , 2008. We do not have information on the evolution of reduced VAT rates across goods over time, but we do not expect those to have changed much as the standard VAT rates changed very little during the time period we consider. We then compute the log of one plus the average across partners of the VAT rates that apply in each country and industry. *Source: European Commission (2008).*

Productivity Real labor productivity is value added divided by the number of employees at the 4-digit Nace Rev.1 level, deflated by GDP deflators. We then use the log of the average across partners of real labor productivity. *Source: Eurostat's New Cronos database.*

Capital Shares Capital shares are value added minus personnel costs, divided by personnel costs. We then calculate the log of one plus the absolute difference in capital shares between countries for each industry and in each year. *Source: Eurostat's New Cronos database.*

Number of Product Categories Source: Eurostat's Prodcoms.

GDP Deflators Value of GDP (million Euros) and volume of GDP (millions of 1995 constant Euros). The deflator is computed as the ratio between the value and the volume of GDP. *Source: Eurostat's New Cronos database.*

CESifo Working Paper Series

for full list see www.cesifo-group.org/wp (address: Poschingerstr. 5, 81679 Munich, Germany, office@cesifo.de)

- 2533 Erkki Koskela and Jan König, The Role of Profit Sharing in a Dual Labour Market with Flexible Outsourcing, January 2009
- 2534 Tomasz Michalak, Jacob Engwerda and Joseph Plasmans, Strategic Interactions between Fiscal and Monetary Authorities in a Multi-Country New-Keynesian Model of a Monetary Union, January 2009
- 2535 Michael Overesch and Johannes Rincke, What Drives Corporate Tax Rates Down? A Reassessment of Globalization, Tax Competition, and Dynamic Adjustment to Shocks, February 2009
- 2536 Xenia Matschke and Anja Schöttner, Antidumping as Strategic Trade Policy Under Asymmetric Information, February 2009
- 2537 John Whalley, Weimin Zhou and Xiaopeng An, Chinese Experience with Global 3G Standard-Setting, February 2009
- 2538 Claus Thustrup Kreiner and Nicolaj Verdelin, Optimal Provision of Public Goods: A Synthesis, February 2009
- 2539 Jerome L. Stein, Application of Stochastic Optimal Control to Financial Market Debt Crises, February 2009
- 2540 Lars P. Feld and Jost H. Heckemeyer, FDI and Taxation: A Meta-Study, February 2009
- 2541 Philipp C. Bauer and Regina T. Riphahn, Age at School Entry and Intergenerational Educational Mobility, February 2009
- 2542 Thomas Eichner and Rüdiger Pethig, Carbon Leakage, the Green Paradox and Perfect Future Markets, February 2009
- 2543 M. Hashem Pesaran, Andreas Pick and Allan Timmermann, Variable Selection and Inference for Multi-period Forecasting Problems, February 2009
- 2544 Mathias Hoffmann and Iryna Shcherbakova, Consumption Risk Sharing over the Business Cycle: the Role of Small Firms' Access to Credit Markets, February 2009
- 2545 John Beirne, Guglielmo Maria Caporale, Marianne Schulze-Ghattas and Nicola Spagnolo, Volatility Spillovers and Contagion from Mature to Emerging Stock Markets, February 2009
- 2546 Ali Bayar and Bram Smeets, Economic and Political Determinants of Budget Deficits in the European Union: A Dynamic Random Coefficient Approach, February 2009

- 2547 Jan K. Brueckner and Anming Zhang, Airline Emission Charges: Effects on Airfares, Service Quality, and Aircraft Design, February 2009
- 2548 Dolores Messer and Stefan C. Wolter, Money Matters Evidence from a Large-Scale Randomized Field Experiment with Vouchers for Adult Training, February 2009
- 2549 Johannes Rincke and Christian Traxler, Deterrence through Word of Mouth, February 2009
- 2550 Gabriella Legrenzi, Asymmetric and Non-Linear Adjustments in Local Fiscal Policy, February 2009
- 2551 Bruno S. Frey, David A. Savage and Benno Torgler, Surviving the Titanic Disaster: Economic, Natural and Social Determinants, February 2009
- 2552 Per Engström, Patrik Hesselius and Bertil Holmlund, Vacancy Referrals, Job Search, and the Duration of Unemployment: A Randomized Experiment, February 2009
- 2553 Giorgio Bellettini, Carlotta Berti Ceroni and Giovanni Prarolo, Political Persistence, Connections and Economic Growth, February 2009
- 2554 Steinar Holden and Fredrik Wulfsberg, Wage Rigidity, Institutions, and Inflation, February 2009
- 2555 Alexander Haupt and Tim Krieger, The Role of Mobility in Tax and Subsidy Competition, February 2009
- 2556 Harald Badinger and Peter Egger, Estimation of Higher-Order Spatial Autoregressive Panel Data Error Component Models, February 2009
- 2557 Christian Keuschnigg, Corporate Taxation and the Welfare State, February 2009
- 2558 Marcel Gérard, Hubert Jayet and Sonia Paty, Tax Interactions among Belgian Municipalities: Does Language Matter?, February 2009
- 2559 António Afonso and Christophe Rault, Budgetary and External Imbalances Relationship: A Panel Data Diagnostic, February 2009
- 2560 Stefan Krasa and Mattias Polborn, Political Competition between Differentiated Candidates, February 2009
- 2561 Carsten Hefeker, Taxation, Corruption and the Exchange Rate Regime, February 2009
- 2562 Jiahua Che and Gerald Willmann, The Economics of a Multilateral Investment Agreement, February 2009
- 2563 Scott Alan Carson, Demographic, Residential, and Socioeconomic Effects on the Distribution of 19th Century US White Statures, February 2009

- 2564 Philipp Harms, Oliver Lorz and Dieter Urban, Offshoring along the Production Chain, February 2009
- 2565 Patricia Apps, Ngo Van Long and Ray Rees, Optimal Piecewise Linear Income Taxation, February 2009
- 2566 John Whalley and Shunming Zhang, On the Arbitrariness of Consumption, February 2009
- 2567 Marie-Louise Leroux, Endogenous Differential Mortality, Non-Contractible Effort and Non Linear Taxation, March 2009
- 2568 Joanna Bęza-Bojanowska and Ronald MacDonald, The Behavioural Zloty/Euro Equilibrium Exchange Rate, March 2009
- 2569 Bart Cockx and Matteo Picchio, Are Short-Lived Jobs Stepping Stones to Long-Lasting Jobs?, March 2009
- 2570 David Card, Jochen Kluve and Andrea Weber, Active Labor Market Policy Evaluations: A Meta-analysis, March 2009
- 2571 Frederick van der Ploeg and Anthony J. Venables, Harnessing Windfall Revenues: Optimal Policies for Resource-Rich Developing Economies, March 2009
- 2572 Ondřej Schneider, Reforming Pensions in Europe: Economic Fundamentals and Political Factors, March 2009
- 2573 Jo Thori Lind, Karl Ove Moene and Fredrik Willumsen, Opium for the Masses? Conflict-Induced Narcotics Production in Afghanistan, March 2009
- 2574 Silvia Marchesi, Laura Sabani and Axel Dreher, Agency and Communication in IMF Conditional Lending: Theory and Empirical Evidence, March 2009
- 2575 Carlo Altavilla and Matteo Ciccarelli, The Effects of Monetary Policy on Unemployment Dynamics under Model Uncertainty - Evidence from the US and the Euro Area, March 2009
- 2576 Falko Fecht, Kjell G. Nyborg and Jörg Rocholl, The Price of Liquidity: Bank Characteristics and Market Conditions, March 2009
- 2577 Giorgio Bellettini and Filippo Taddei, Real Estate Prices and the Importance of Bequest Taxation, March 2009
- 2578 Annette Bergemann and Regina T. Riphahn, Female Labor Supply and Parental Leave Benefits – The Causal Effect of Paying Higher Transfers for a Shorter Period of Time, March 2009
- 2579 Thomas Eichner and Rüdiger Pethig, EU-Type Carbon Emissions Trade and the Distributional Impact of Overlapping Emissions Taxes, March 2009

- 2580 Antonios Antypas, Guglielmo Maria Caporale, Nikolaos Kourogenis and Nikitas Pittis, Selectivity, Market Timing and the Morningstar Star-Rating System, March 2009
- 2581 António Afonso and Christophe Rault, Bootstrap Panel Granger-Causality between Government Budget and External Deficits for the EU, March 2009
- 2582 Bernd Süssmuth, Malte Heyne and Wolfgang Maennig, Induced Civic Pride and Integration, March 2009
- 2583 Martin Peitz and Markus Reisinger, Indirect Taxation in Vertical Oligopoly, March 2009
- 2584 Petra M. Geraats, Trends in Monetary Policy Transparency, March 2009
- 2585 Johannes Abeler, Armin Falk, Lorenz Götte and David Huffman, Reference Points and Effort Provision, March 2009
- 2586 Wolfram F. Richter, Taxing Education in Ramsey's Tradition, March 2009
- 2587 Yin-Wong Cheung, Menzie D. Chinn and Eiji Fujii, China's Current Account and Exchange Rate, March 2009
- 2588 Alexander Haupt and Silke Uebelmesser, Voting on Labour-Market Integration and Education Policy when Citizens Differ in Mobility and Ability, March 2009
- 2589 Hans Jarle Kind, Marko Koethenbuerger and Guttorm Schjelderup, Should Utility-Reducing Media Advertising be Taxed?, March 2009
- 2590 Alessandro Cigno, How to Avoid a Pension Crisis: A Question of Intelligent System Design, March 2009
- 2591 Helmut Lütkepohl and Fang Xu, The Role of the Log Transformation in Forecasting Economic Variables, March 2009
- 2592 Rainald Borck, Hyun-Ju Koh and Michael Pflüger, Inefficient Lock-in and Subsidy Competition, March 2009
- 2593 Paolo M. Panteghini, On the Equivalence between Labor and Consumption Taxation, March 2009
- 2594 Bruno S. Frey, Economists in the PITS?, March 2009
- 2595 Natalie Chen and Dennis Novy, International Trade Integration: A Disaggregated Approach, March 2009